University of Rajshahi	Rajshahi-6205	Bangladesh.
RUCL Institutional Repository		http://rulrepository.ru.ac.bd
Department of Mathematics		PhD Thesis

1993

Quantum Gravity in Curved Space-Time of General Relativity

Mondal, Asit Kumar

University of Rajshahi

http://rulrepository.ru.ac.bd/handle/123456789/885 Copyright to the University of Rajshahi. All rights reserved. Downloaded from RUCL Institutional Repository. " Symmetry , as wide or as narrow as you may difine it , is one idea by which man through the ages has tried to comprehend and create order, beauty and perfection . "

- Hermann Weyl

" Beauty is the proper conformity of the parts to one another and to the whole . "

- Werner Heisenberg

QUANTUM GRAVITY IN CURVED SPACE-TIME OF GENERAL RELATIVITY

BY

ASIT KUMAR MONDAL

A thesis submitted for the degree of Doctor of Philosophy in the University of Rajshahi

Department of Mathematics University of Rajshahi Rajshahi - 6205 BANGLADESH 1993

Rojshabi Uak ereby Libaky Documentati a kashon Documentati Documen

DEDICATED TO MY PARENTS

D-2332 Doc 510 MOQ

ACKN OWLE DGMEN TS

I wish to express my deep gratitude to my supervisor Professor Mainuddin Ahmed , for his guidance and encouragement throughout the course of this work and in the preparation of this thesis .

I am grateful to the University Grants Commission for awarding a fellowship (Jan 1988 - Dec 1991) and the financial assistance to wards the preparation of the thesis.

I am also grateful to the Ministry of Education , Bangladesh for granting deputation / study leave during the tenure of the U G C fellowship .

My regards are also due to all of my teachers, Department of Mathematics, Rajshahi University and colleagues especially to our Departmental Head Prof. A.B. Mitra for cooperation and inspiration that I have had from them during this work .

Many thanks go to my wife Ratna for support and encouragement during the entire period of this work . Thanks are also due to Mr. T.K. Dey for sending some rare and important papers from abroad .

Finally, I would like to thank Mr. M. Hoque for his patient typing and retyping the manuscript.

NOTATIONS AND CONVENTIONS

Bar indicates complex conjugation. Dot denotes differentiation with respect to the argument.

Units are choosen such that G = c = h = 1, where G is the Newtonian gravitational constant, c is the speed of light and h is the Planck's constant.

We shall adopt the following abbreviations :

Newman - Unti - Tamburino	NUT
NUT - Kerr - Newman	NUTKN
NUT - Kerr - Newman - de Sitter	HNU TKN
NUT - Kerr - Newman - Kasuya	NU TKNK
Kerr - Newman - Kasuya - de Sitter	HKNK
Kerr - Newman - de Sitter	HKN

ABSTRACT

The main contents of this thesis may be briefly summarized as follows :

Chapter 1 contains a brief account of the Plebanski spacetime. We specify there the various special class of spacetimes covered by the Plebanski spacetime. Among the various special cases of the Plebanski spacetime a new spacetime discovered there is the NUT - Kerr - Newman - Kasuya - de Sitter spacetime. This chapter also includes a discussion on the modification of the Plebanski spacetime and presents the modified form of the Plebanski spacetime.

Chapter 2 deals with the separation of the variables of the Dirac equation in an arbitrary curved background spacetime. We discuss there some of the special cases of the separated Dirac equation. The pertinent equation of the separated Dirac equation will be used to derive the radial decoupled Dirac equations for the concerned background spacetimes which will be useful in studying the problems of Hawking radiation.

Since our efforts are concentrated on quantum field theory in some interesting spacetimes of general relativity which are not the black hole spacetimes but include the black hole spacetimes as special cases , we review in Chapter 3 , Hawking's quantum effects near the event horizon of NUT-Kerr-Newman spacetime containing flat black hole spacetimes as special cases .

Chapter 4 also reviews and extends the result obtained in Chapter 3. The Hawking radiation of Dirac particles is studied in NUT-Kerr-Newman de Sitter spacetime containing black hole spacetimes which are asymptotically flat as well as asymptotically de Sitter as special cases.

Chapter 5 includes the investigation of Dirac particles in Kasner-type spacetime .

Chapter 6 presents Hawking's thermal radiation by black hole near the horizons of the Plebanski spacetime containing a large number of spacetimes of which some are important from the physical point of view. The result obtained in this chapter not only encompasses the known results but also includes some new results. This chapter ends with a concluding remarks.

CON TEN TS

-: vii :-

IN TRODUCTION		Page No. 1
CHAPTER 1 - PLEBANSKI SPACETIME		7
1.1 Some P	roperties of the Plebanski Spacetime	8
1.2 Contrac	ctions of the Plebanski Spacetime	10
1.3 Canoni	cal forms of the Contracted Solutions	16
1.4 The Cor	nbined NUT-Kerr-Newman-Kasuya Spacetime	24
1.5 The Con Spacet	mbined NUT-Kerr-Newman-Kasuya de Sitter ime	27
1.6 A Modi	fied form of the Plebanski Spacetime	30
	SEPARATED DIRAC EQUATION IN AN ARBITRARY CURVED SPACETIME	34
2.1 Arbitr	ary Curved Spacetime and Some Preliminaries	35
2.2 Separa	tion of the Dirac Equation	38
2.3 Reduct	ion to Special Cases	48
CHAPTER 3 -	HAWKING RADIATION OF DIRAC PARTICLES IN NUT-KERR-NEWMAN SPACETIME	53
3.1 The N	UTKN Spacetime	54
3.2 Radial	Wave Equation	55
3.3 Hawkin	g Thermal Spectrum	58
3.4 Discus	sion	62

-: V111 :-

-		Page No.
CHAPT	ER 4 – HAWKING RADIATION OF DIRAC PARTICLES IN NUT-KERR-NEWMAN – DE SITTER SPACETIME	64
4.1	The HNUTKN Spacetime	65
4.2	Radial Wave Equation	66
4.3	Hawking Thermal Spectrum	70
4.4	Discussion	75
CHAPT	ER 5 - HAWKING RADIATION OF DIRAC PARTICLES IN KASNER - TYPE SPACETIME	76
5.1	The Kasner Type Spacetime	77
5.2	Radial Wave Equation	78
5.3	Hawking Thermal Spectrum	79
5.4	Remarks	83
CHAPT	ER 6 – HAWKING RADIATION OF DIRAC PARTICLES IN PLEBANSKI SPACETIME	85
6.1	The Plebanski Spacetime	85
6.2	Radial Wave Equation	86
6.3	Hawking Thermal Spectrum	90
6.4	Discussion	95

RE FE REN CE S

~

.

96

INTRODUCTION

The problem of reconciling the description of the gravitational force embodied in Einstein's general theory of relativity with quantum theory is a central issue in theoretical physics. There are two approaches in which quantum idea can be brought into relativity. In the dynamic approach one requires the investigation of quantum equivalent of Einstein's equation. Unfortunately there are some difficulties in this approach to quantize gravity itself. However, there is a great deal of work going in this area and some progress has been made.

In the kinematic approach one tries to study the quantized matter fields in the presence of the gravitational field. For example, writing the Dirac equation in curved spacetime, one studies how Dirac particles behave in a gravitational field such as that of a black hole. The limitation of this approach is that it does not help us to quantize the gravitational field itself. In spite of the limitations, one could hope that the results obtained in this semi-classical approach may be an integral part of the full quantum gravity and the gravitational field will be quantized.

Actually, the investigation of matter fields in curved spacetime (Friedman - Robertson - Walker) was initiated by Schrodinger [1] in 1939, although Rosenfeld [2] seems to be the first to treat the quantum mechanical interaction of matter and gravity. Following the original lead of Schrodinger, in the late sixtles Parker [3] and Sexl and Urbantke [4] began discussion of the creation of particles during the initial rapidly expanding phase of the universe. However, the investigation of matter fields in curved spacetime caught fire when Hawking [5] in 1974, demonstrated how a black hole can in fact lose mass by a process of quantum particle creation. Hawking's quantum effects interpreted as the emission of a thermal spectrum of particles near black hole event horizon, opened a new and existing era in quantum gravity research.

Hawking [6] used the technique of quantum field theory on a given background spacetime and showed that black holes create and emit particles at a steady rate and that the predicted rate is just that of the thermal emission of a body with the temperature $\frac{K}{2\pi}$, where K is the surface gravity [7] of the black hole.

Gibbons and Hawking [8] extended Hawking's [9] results on thermal radiation by black holes near black hole event horizon to the cosmological event horizon and deduced the thermodynamic properties of black holes using the arguments of Bekenstein [10]. Subsequently, several studies that attempt to relate the thermal properties of black holes to quantum gravity have appeared [11,12].

Hawking's result on thermal radiation by black holes have been studied by different authors in different types of spacetimes such as Kerr [13], Kerr-Newman [14], Kerr-Newman-de Sitter [15,16] spacetimes. Recently Ahmed [17,18] studied Hawking radiation of Dirac particles near the horizons of NUT-Kerr-Newman and NUT-Kerr-Newman - de Sitter spacetimes. Ahmed's works are interesting in that Hawking's and Gibbons and Hawking's results could also be obtained

in the case of NUT and NUT - de Sitter spacetimes which have very peculiar properties. Most recently Ahmed and Mondal [19] have shown that Hawking's result on thermal radiation by black holes also occurs in the case of Kasner - type spacetime. In this work Ahmed and Mondal observed that particle is creating in Kasner spacetime which is contraction phase of the Schwarzschild spacetime. So we have seen that the particles are creating in the contraction phase of the universe which is in contradiction to the fact that particles are created in the expanding phase of the universe. Particles should be disappeared in the contraction phase of the universe whereas they are being created. To overcome the awkward situation we have interpreted the result as saying that souls of the particles are creating in the contraction phase of the universe.

From the works of Ahmed [20] and Ahmed and Mondal [21] it is clear that Hawking's result could be obtained in the case of spacetimes which are not black hole spacetimes. The aim of this thesis is to show that Hawking's and Gibbons and Hawking's results could also be obtained in a more general background spacetime viz. Plebanski spacetime having horizons.

-: 3 :-

In Chapter 1 , the various special cases of the Plebanski spacetime have been described. It has been shown that the Plebanski spacetime includes a large number of solutions of Einstein or Einstein-Maxwell equations with or without cosmological constant. Besides the black hole spacetimes (asymptotically flat as well as asymptotically de Sitter '), the Plebanski spacetime includes many interesting spacetimes which are not black hole spacetimes but are important from the physical point of view . While most of the various special cases of the Plebanski spacetime have been discussed in the literature [22], a number of new spacetimes (such as Kerr - de Sitter , NUT - Kerr de Sitter, NUT - Kerr - Newman - Kasuya - de Sitter spacetimes and so forth) do not appear to have been stated explicitly in the literature previously. It is shown that with appropriate specialization of the parameters and coordinate transformation the NUT - Kerr - Newman -Kasuya - de Sitter spacetime can be derived from the Plebanski spacetime . It is also shown that if the Plebanski spacetime is modified then the modified form of the Plebanski spacetime yields the various special cases of the Plebanski spacetime in an elegant way .

As we would like to study Hawking radiation of Dirac particles in curved spacetime, it is necessary to separate the Dirac equation in the background spacetime considered. Kamran and McLenaghan [23] separated Dirac equation in an arbitrary curved background spacetime and obtained the decoupled quantum field equations. The second chapter is devoted to study of the separation of variables of Dirac equation and to show that the earlier results of the separated Dirac equation of Chandrasekhar[24,25] and Page [26] are included in the present general result.

Taking the advantages of the separation of Dirac equation in an arbitrary curved background spacetime we derive the radial decoupled Dirac equations in the NUT-Kerr - Newman , NUT-Kerr - Newmande Sitter , Kasner - type and Plebanski background spacetimes in a limiting procedure and then study Hawking radiation in Chapters 3, 4,5 and 6.

Chapter 3 reviews the Hawking radiation of Dirac particles near the event horizon of NUT - Kerr- Newman spacetime which includes the asymptotically flat black hole spacetimes as special cases. From this result it is clear that Hawking's result could be obtained in the case of NUT spacetime.

Chapter 4 also reviews the Hawking radiation of Dirac particles near the event horizon and cosmological horizon of NUT - Kerr -Newman - de Sitter spacetime which includes the black hole spacetimes which are asymptotically flat as well as asymptotically de Sitter as special cases. Here it is observed that Hawking's result is also valid in the case of NUT - de Sitter spacetime.

Chapter 5 includes our study of Dirac particles in Kasner type spacetime which gives very interesting results that particles are created in the contraction phase of the universe.

-: 5 :-

Chapter 6 presents the Hawking radiation of Dirac particles near the horizons of the Plebanski spacetime. The result obtained in this chapter includes the results of Chapters 3, 4 and 5. This result not only encompasses the results obtained in [27] but also includes some new results. The last section of this chapter ends with conclusive remarks. -: 7 :-

CHAPTER - 1

PLEBANSKI SPACETIME

Plebanski [28] studied a class of solutions of Einstein-Maxwell equations. In Boyer coordinates (p, σ , q, τ) the Plebanski spacetime is given by

$$ds^{2} = \frac{p^{2} + q^{2}}{x} dp^{2} + \frac{x}{p^{2} + q^{2}} (d\tau + q^{2} d\sigma)^{2} + \frac{p^{2} + q^{2}}{y} dq^{2}$$
$$- \frac{y}{p^{2} + q^{2}} (d\tau - p^{2} d\sigma)^{2} \qquad \dots \quad (1.1a)$$

where

$$X = X(p) = b - g^{2} + 2np - \varepsilon p^{2} - (\Lambda/3) p^{4} \qquad (1.1b)$$

$$Y = Y(q) = b + e^{2} - 2Mq + \varepsilon q^{2} - (\Lambda/3) q^{4} \qquad (1.1c)$$

with electric potential

$$A_{\mu} dx^{\mu} = \frac{eq}{p^{2} + q^{2}} (d\tau - p^{2} d\sigma) \qquad (1.2)$$

Besides the cosmological constant \wedge , the metric (1.1) contains six parameters b, e, g, M, n, ε . Under the proper coordinate transformation along with the suitable adjustment of the kinetical parameters b and ε , the metric (1.1) gives many physically interesting solutions of Einstein or Einstein - Maxwell equations with or without cosmological constant. The surfaces Y = 0 have been interpreted as the horizon of the metric.

1.1 Some Properties of the Plebanski Spacetime

The electromagnetic field associated with the metric is given by

$$W = -d \left\{ \frac{e + ig}{q + ip} (d_{\tau} - i p q d_{\sigma}) \right\} \qquad \dots (1.3)$$

The components of the Weyl tensor $C^{(a)}$ are

$$C^{(1)} = C^{(2)} = C^{(4)} = C^{(5)} = 0$$

$$\dots (1.4)$$

$$C^{(3)} = \frac{1}{6} \frac{\ddot{x} + \ddot{y}}{p^{2} + q^{2}} - \frac{1}{p^{2} + q^{2}} \left\{ \frac{\ddot{x} + i\ddot{x}}{q + ip} - \frac{2(Y - X)}{(q + 1p)^{2}} \right\}$$

The components of Ricci tensor R_{ab} are

$$R_{11} = R_{22} = R_{13} = R_{33} = R_{14} = R_{44} = R_{23} = R_{24} = 0$$

$$R_{12} = \frac{1}{2} \frac{\ddot{x}}{p^{2} + q^{2}} - \frac{1}{p^{2} + q^{2}} \left\{ (p \dot{x} - q \dot{y}) - (x - y) \right\}$$

$$R_{34} = \frac{1}{2} \frac{\ddot{y}}{p^{2} + q^{2}} + \frac{1}{p^{2} + q^{2}} \left\{ (p \dot{x} - q \dot{y}) - (x - y) \right\}$$

-: 9 :-

The scalar curvature R is

$$R = \frac{\ddot{x} + \ddot{y}}{p^2 + q^2} = 2(R_{12} + R_{34}) \qquad \dots \qquad (1.6)$$

where here and above dots denote differentiation with respect to the argument.

Now we turn to the equation (1.4). This equation can be simplified further for $C^{(3)}$. Using equations (1.1b) and (1.1c) we obtain from (1.4)

$$C^{(3)} = \frac{-2}{(q^2 + p^2)(q + ip)^2} [Mq + np - e^2 - g^2 - i(Mp - nq)] \qquad (1.7)$$

Consequently if any of the constants M,n,e,g is not zero, the metric (1.1) is of Petrov type D. In the case $\begin{pmatrix} (3) \\ = 0 \end{pmatrix}$, the metric is conformally flat. In addition to $C^{(a)} = 0$, if also $R_{ab} = 0$, then the metric becomes flat. The metric is asymptotically de Sitter when $\wedge \neq 0$ but asymptotically flat when $\wedge = 0$.

A large number of solutions can be obtained by contracting the metric (1.1) by appropriate limiting procedures. In the course of this discussion, we give the interpretation of the parameters involved in the metric (1.1) and their correspondence to some well-known solutions.

-: 10 :-

1.2 Contractions of the Plebanski Spacetime

(a) We consider the coordinate transformation from $x^{\mu} = (p, \sigma, q, \tau)$ to $x'^{\mu} = (p', \sigma', q', \tau')$ defined by

$$p = p_{o} + \varepsilon_{o} p', \sigma = \sigma / \varepsilon_{o}, q = q', \tau = \tau' + (p_{o}^{2} \sigma') / \varepsilon_{o} \qquad \dots \qquad (1.8)$$

where p_0 is an arbitrary constant and ε_0 denotes the contraction parameter .

Under the coordinate transformation given by (1.8) the metric (1.1) reduces to the form

$$ds^{2} = \frac{\{(p_{0} + \epsilon_{0} p')^{2} + q'^{2}\}}{\epsilon_{0}^{-2} \chi (p_{0} + \epsilon_{0} p')} dp'^{2}}$$

$$+ \frac{\epsilon_{0}^{-2} \chi (p_{0} + \epsilon_{0} p')}{\{(p_{0} + \epsilon_{0} p')^{2} + q'^{2}\}} \{\epsilon_{0} d\tau' + (p_{0}^{2} + q'^{2}) d\sigma'\}^{2}$$

$$+ \frac{1}{\{(p_{0} + \epsilon_{0} p')^{2} + q'^{2}\}^{-1} \chi (q')} dq'^{2}$$

$$- \{(p_{0} + \epsilon_{0} p')^{2} + q'^{2}\}^{-1} \chi (q') \{d\tau' - (2p_{0} p' + \epsilon_{0} p'^{2}) d\sigma'\}^{2} (1.9a)$$

where

$$\frac{-2}{\varepsilon_{o}} \chi \left(p_{o} + \varepsilon_{o} p' \right) = \varepsilon_{o}^{-2} \left\{ b - g^{2} + 2n \left(p_{o} + \varepsilon_{o} p' \right) - \varepsilon \left(p_{o} + \varepsilon_{o} p' \right)^{2} - \left(\frac{\sqrt{3}}{9} \right) \left(p_{o} + \varepsilon_{o} p' \right)^{4} \right\} \dots (1.9b)$$

-: 11 :-

$$\{(p_{0} + \varepsilon_{0} p')^{2} + q'^{2}\}^{-1} \Upsilon(q')$$

$$= \{(p_{o} + \varepsilon_{o} p')^{2} + q'\}^{-1}(b + e^{2} - 2Mq' + \varepsilon q'^{2} - \frac{\Lambda}{-}q'^{4}) \quad (1.9c)$$

Simplifying we can write equation (1.9) in the form

$$ds^{2} = \frac{\{(p_{0} + \varepsilon_{0} p')^{2} + q'^{2}\}}{x_{1}} dp'^{2} + \frac{x_{1}}{\{(p_{0} + \varepsilon_{0} p')^{2} + q'^{2}\}} \varepsilon_{0} d\tau' + (p_{0}^{2} + q'^{2}) d\sigma'\}^{2}}{(p_{0} + \varepsilon_{0} p')^{2} + q'^{2}\}} + \frac{dq'^{2}}{y_{1}} Y_{1} \{d\tau' - (2p_{0} p' + \varepsilon_{0} p'^{2}) d\sigma'\}^{2} \dots (1.10a)$$

where

 $\gamma_{o} = \epsilon + 2 \wedge p_{o}^{2}$

$$X_{1} = \varepsilon_{0}^{-2} X (p_{0} + \varepsilon_{0} p') = \alpha_{0} + 2\beta_{0}p' - \gamma_{0} p'^{2} - \frac{4}{3} \wedge p_{0} \varepsilon_{0} p'^{3} - \frac{\Lambda}{3} \varepsilon_{0}^{2} p'^{4} (1.10b)$$
$$Y_{1} = \{ (p_{0} + \varepsilon_{0} p')^{2} + q'^{2} \}^{-1} Y (q')$$

$$= \{ (p_0 + \varepsilon_0 p')^2 + q'^2 \} (a_0 \varepsilon_0^2 - 2\beta_0 p_0 \varepsilon_0 - \gamma_0 p_0^2 + g^2 + \Lambda p_0^4 \}$$

$$e^2 - 2Mq' + \epsilon q'^2 - \frac{\Lambda}{-} q'^4)$$
 ... (1.10c)

$$a_{o} = \epsilon_{o}^{-2} (b - g^{2} + 2np_{o} - \epsilon p_{o} - \frac{\Lambda}{3} p_{o}^{4})$$
 (1.10d)

 $\beta_{0} = \varepsilon_{0}^{-2} (n - \varepsilon p_{0} - \frac{2 \wedge}{3} p_{0}^{3}) \qquad (1.10e)$

. . (1.10f)

Rajshahi University Library Documentation Section Document No D-2332 Date 24, 10, 04 with α_{o} , β_{o} and γ_{o} being constants independent on ϵ_{o} .

Now if $\varepsilon_0 \rightarrow 0$, then we have from (1.10)

$$ds^{2} = (p_{0}^{2} + q'^{2}) (\frac{dp'^{2}}{x_{1}} + x_{1} d\sigma'^{2}) + \frac{dq'^{2}}{x_{1}} - y_{1} (dr' - 2p_{0}p'd\sigma')^{2} \dots (1.1a)$$

where

$$X_{1} = \alpha_{0} + 2\beta_{0} p' - \gamma_{0} p'^{2} \qquad \dots \qquad (1.11b)$$

$$Y_{1} = (p_{0}^{2} + q'^{2}) (g^{2} - \gamma_{0} p_{0}^{2} + \wedge p_{0}^{4} + e^{2} - 2Mq' + eq'^{2} - \frac{\Lambda}{3}q'^{4}) \qquad \dots \qquad (1.11c)$$

The metric (1.11) can ultimately be put in the form

$$ds^{2} = (p_{0}^{2} + q'^{2}) (\frac{dp'^{2}}{x_{1}} + x_{1} d\sigma'^{2}) + \frac{dq'^{2}}{x_{1}}$$
$$- x_{1} (d\tau' - 2p_{0}p'd\sigma')^{2} \qquad \dots (1.12a)$$

where

$$X_{1} = \alpha_{0} + 2 \beta_{0} p' - \gamma_{0} p'^{2}$$
 (1.12b)

$$Y_{1} = \gamma_{0} - \frac{\Lambda}{3} (q'^{2} + 5 p_{0}^{2}) - 2 \operatorname{Re} \left(\frac{M + \ln_{0}}{q' + 1 p_{0}}\right) + \left|\frac{e + 1g}{q' + 1 p_{0}}\right|^{2}$$
(1.12c)
$$n_{0} = \gamma_{0} p_{0} - \frac{4\Lambda}{3} p_{0}^{3}$$
(1.12d)

The contracted solution given by equation (1.12) in the coordinates $x'^{\mu} = (p', \sigma', q', \tau')$ is the generalized NUT solution.

(b) We now consider the coordinate transformation from $\mathbf{x}^{\mu} = (p, \sigma, q, \tau)$ to $\mathbf{x}'^{\mu} = (p', \sigma', q', \tau')$ defined by

$$p = p', \sigma = \sigma' / \varepsilon_{\sigma}, q = q_{\sigma} + \varepsilon_{\sigma} q', \tau = \tau' - (q_{\sigma}^2 \sigma') / \varepsilon_{\sigma} \dots (1.13)$$

where q_0 is arbitrary constant and ε_0 is the contraction parameter .

Under the coordinate transformation (1.13) the metric (1.1) reduces to the form

$$ds^{2} = \frac{1}{\{p'^{2} + (q_{0} + \varepsilon_{0}q')^{2}\}^{-1} X(p')} dp'^{2}}$$

$$+ \{p'^{2} + (q_{0} + \varepsilon_{0}q')^{2}\}^{-1} X(p') \{d\tau' + (2q_{0}q' + \varepsilon_{0}q'^{2})dd'\}^{2}$$

$$+ \frac{\{p'^{2} + (q_{0} + \varepsilon_{0}q')^{2}\}}{\varepsilon_{0}^{-2} Y(q_{0} + \varepsilon_{0}q')} dq'^{2}$$

$$- \frac{\varepsilon_{0}^{-2} Y(q_{0} + \varepsilon_{0}q')}{\{p'^{2} + (q_{0} + \varepsilon_{0}q')^{2}\}} \{\varepsilon_{0} d\tau' - (q_{0}^{2} + p'^{2}) dd'\}^{2} \dots (1.14a)$$

-: 14 :-

where

$$\{ p'^{2} + (q_{0} + \epsilon_{0}q')^{2} \}^{-1} X (p')$$

$$= \{ p'^{2} + (q_{0} + \epsilon_{0}q')^{2} \}^{-1} (b - g^{2} + 2n p' - \epsilon p'^{2} - \frac{h}{3} p'^{4}) \qquad (1.14b)$$

$$\epsilon_{0}^{-2} Y (q_{0} + \epsilon_{0}q')$$

$$= \epsilon_{0}^{-2} \{ b + e^{2} - 2M (q_{0} + \epsilon_{0}q') + \epsilon (q_{0} + \epsilon_{0}q')^{2} - \frac{h}{3} (q_{0} + \epsilon_{0}q)^{4} \} (1.14c)$$
Simplifying we can write equation (1.14) in the form

$$ds^{2} = \frac{dp'^{2}}{X_{2}} + X_{2} \{ d\tau' + (2q_{0}q') + \epsilon_{0}q'^{2} \} ds' \}^{2} + \frac{p'^{2} + (q_{0} + \epsilon_{0}q')^{2}}{Y_{2}} dq'^{2} + \frac{p'^{2} + (q_{0} + \epsilon_{0}q')^{2}}{Y_{2}} dq'^{2} + \frac{1}{p'^{2} + (q_{0} + \epsilon_{0}q')^{2}} \{ \epsilon_{0}d\tau' - (q_{0}^{2} + p'^{2}) ds' \}^{2} \dots (1.15a)$$

where

•

$$Y_{2} = \varepsilon_{0}^{-2} Y (q_{0} + \varepsilon_{0} q')$$

= $\alpha_{1} + 2\beta_{1} q' + \gamma_{1} q'^{2} - \frac{4}{3} \wedge q_{0} q' \varepsilon_{0} - \frac{1}{3} \wedge q'^{4} \varepsilon_{0}^{2} \cdots (1.15b)$

.

-: 15 :-

3

•

$$\begin{split} x_{2} &= \left\{ p^{2} + (q_{0} + \epsilon_{0}q^{\prime})^{2} \right\}^{-1} \chi (p^{\prime}) \\ &= \left\{ p^{\prime 2} + (q_{0} + \epsilon_{0}q^{\prime})^{2} \right\}^{-1} \left\{ a_{1} \epsilon_{0}^{2} - 2\beta_{1} q_{0} \epsilon_{0} + \gamma_{0} q_{0}^{2} + \wedge q_{0}^{4} \\ &- e^{2} - g^{2} + 2np^{\prime} - \epsilon p^{\prime}^{2} - \frac{1}{3} \wedge p^{\prime}^{4} \right\} \quad \dots \quad (1.15c) \\ a_{1} &= \epsilon_{0}^{-2} (b + e^{2} - 2Mq_{0} + \epsilon q_{0}^{2} - \frac{1}{3} \wedge q_{0}^{4}) \qquad \dots \quad (1.15d) \\ \beta_{1} &= \epsilon_{0}^{-1} (\epsilon q_{0} - M - \frac{2}{3} \wedge q_{0}^{3}) \qquad \dots \quad (1.15e) \\ \gamma_{1} &= \epsilon - 2 \wedge q_{0}^{2} \qquad \dots \quad (1.15f) \\ \gamma_{1} &= \epsilon - 2 \wedge q_{0}^{2} \qquad \dots \quad (1.15f) \\ \gamma_{1} &= \epsilon - 2 \wedge q_{0}^{2} \qquad \dots \quad (1.15f) \\ \gamma_{1} &= \epsilon_{0} \rightarrow 0 , \quad \text{then from } (1.15) \text{ we have} \\ ds^{2} &= \frac{d p^{\prime 2}}{x_{2}} + \chi_{2} (d \tau^{\prime} + 2 q_{0}q^{\prime}d d^{\prime})^{2} \\ &+ (q_{0}^{2} + p^{\prime 2}) (\frac{d q^{\prime 2}}{x_{2}} - \gamma_{2} d d^{\prime 2}) \qquad \dots \quad (1.16e) \\ \gamma_{2} &= a_{1} + 2 \beta_{1}q^{\prime} + \gamma_{1}q^{\prime 2} \qquad \dots \quad (1.16b) \\ \chi_{2} &= (p^{\prime 2} + q_{0}^{2})^{-1} \left\{ \gamma_{0} q_{0}^{2} + \wedge q_{0}^{4} - e^{2} - g^{2} + 2n p^{\prime} + \epsilon p^{\prime 2} - \frac{1}{3} \wedge p^{\prime 4} \right\} \quad (1.16c) \end{split}$$

$$ds^{2} = \frac{d p'^{2}}{X_{2}} + X_{2} (d \tau' + 2 q_{0} q' d \sigma')^{2} + (q_{0}^{2} + p'^{2}) (\frac{d q'^{2}}{Y_{2}} - Y_{2} d \sigma'^{2}) \qquad \dots \qquad (1.17a)$$

$$Y_{2} = \alpha_{1} + 2 \beta_{1} q' + \gamma_{1} q'^{2} \qquad ... (1.17b)$$

$$X_{2} = -\gamma_{1} - \frac{1}{3} \wedge (p'^{2} + 5q_{o}^{2}) + 2 \operatorname{Re}\left(\frac{m_{o} + \operatorname{in}}{q_{o} + \operatorname{i}p'}\right) - \left|\frac{e + \operatorname{ig}}{q_{o} + \operatorname{i}p'}\right|^{2} \quad (1.17c)$$
$$m_{o} = \gamma_{1}q_{o} + \frac{4}{3} \wedge q_{o}^{3} \quad (1.17d)$$

This contracted solution given by (1.17) is the generalized anti - NUT solution .

1.3 <u>Canonical forms of the Contracted Solutions</u>

(a) Let us consider the generalized NUT solution given by (1.12) and restrict the parameter γ_0 associated with it to the discrete values

$$\gamma_0 = 1, 0, -1$$

-: 16 :-

Case I:
$$\gamma_0 = 1$$

If we put $a_0 = 1$, $\beta_0 = 0$ and consider coordinate transformation from $x'^{\mu} = (p', d', q', \tau)$ to $x''^{\mu} = (\theta, \phi, q', t')$ defined by

$$p' = \cos \theta, \ \sigma' = \phi, \ q' = q', \ \tau' = t'$$
 ... (1.18)

then the metric given by (1.12) takes the form

$$ds^{2} = (p_{0}^{2} + q'^{2}) (d\theta^{2} + \sin^{2}\theta d\phi^{2}) + \frac{dq'^{2}}{Y_{1}} - Y_{1} (dt' - 2p_{0}\cos\theta d\phi)^{2} \qquad (1.19a)$$

2

where

$$Y_{1} = 1 - \frac{\Lambda}{3} \left(q'^{2} + 5 p_{0}^{2} \right) - 2 \operatorname{Re} \left(\frac{M + in_{0}}{q' + ip_{0}} \right) + \left| \frac{e + ig}{q' + ip_{0}} \right|^{2} \dots (1.19b)$$

$$n_{0} = p_{0} - \frac{4}{3} \wedge p_{0}^{3}$$
 ... (1.19c)

<u>Case II</u>: $\Upsilon = 0$

If we set $a_0 = 1$, $\beta_0 = 0$ and introduce new coordinates $p' = \theta \cos \phi$, $\sigma' = \theta \sin \phi$, $\tau' = t' + p_0 p' \sigma'$... (1.20) then the metric given by (1.12) reduces to the form -: 18 :-

$$ds^{2} = (p_{0}^{2} + q^{\prime 2})(d\theta^{2} + \theta^{2}d\phi^{2}) + \frac{dq^{\prime 2}}{Y_{1}}$$

- $Y_{1}(dt' + p_{0}\theta^{2}d\phi)^{2}$... (1.21a)

where

$$Y_{1} = -\frac{\Lambda}{3} (q'^{2} + 5p_{0}^{2}) - 2 \operatorname{Re} \left(\frac{M + in_{0}}{q' + ip_{0}} + \frac{e + ig}{q' + ip_{0}}\right)^{2} \dots (1.21b)$$

$$n_{0} = -\frac{4}{3} \wedge p_{0}^{3} \dots (1.21c)$$

Case III:
$$\gamma_0 = -1$$

If we set $a_0 = -1$, $\beta_0 = 0$ and introduce new coordinates $p' = \cosh \theta$, $\sigma' = \phi$, $\tau' = t'$... (1.22)

then we have (1.12) as

$$ds^{2} = (p_{o}^{2} + q'^{2}) (d\theta^{2} + \sin h^{2} \theta d\phi^{2}) + \frac{dq'^{2}}{Y_{1}} - Y_{1} (dt' - 2p_{o} \cosh \theta d\phi)^{2} \qquad \dots (1.23a)$$

where

$$Y_{1} = -1 - \frac{\Lambda}{3} (q'^{2} + 5p_{o}^{2}) - 2Rs \left(\frac{M + in_{o}}{q' + ip_{o}} + \frac{s + ig}{q' + ip_{o}}\right)^{2} \dots (1.23b)$$

$$n_{o} = -p_{o} - \frac{4}{3} \wedge p_{o}^{3}$$
 ... (1.23c)

The equations (1.19), (1.21) and (1.23) are the canonical representations of the generalized NUT solutions.

Now we would like to provide some comments concerning the interpretation of the generalized NUT solution described in canonical forms.

If
$$p_0 = 0$$
, then we have from (1.19)
 $ds^2 = q'^2 (d\theta^2 + \sin^2\theta d\phi^2) + \frac{dq'^2}{y_1} - y_1 dt'^2 \dots (1.24a)$

 $Y_1 = 1 - \frac{\Lambda}{3} q'^2 - \frac{2M}{q'} + \frac{e^2 + g^2}{q'^2}$... (1.24b)

If g = 0 we recognize (1.24) as the Reissner - Nordstrom solution with the cosmological constant \wedge . The coordinate q' plays the role of the radial variable. The constant M and e are interpreted as mass and charge associated with this solution. If g is different from zero, then (1.24) represents a slight generalization of the cosmological Reissner - Nordstrom solution ; g is interpreted as magnetic charge. For e = g = 0 the solution given by (1.24) becomes the Schwarzschild solution generalized by the cosmological constant. Further with $e = g = \Lambda = 0$, we obtain from (1.24), the basic Schwarzschild solution.

If $p \neq 0$ but e = g = 0, then the solution given by (1.19) reduces to the form

$$ds^{2} = (p_{0}^{2} + q'^{2}) (d\theta^{2} + \sin^{2}\theta d\phi^{2}) + \frac{dq'^{2}}{Y_{1}} - Y_{1} (dt' - 2p_{0} \cos\theta d\phi)^{2} \qquad (1.25a)$$

where

$$Y_{1} = 1 - \frac{\Lambda}{3} (q'^{2} + 5p_{0}^{2}) - 2 \operatorname{Re} \left(\frac{M + \ln_{0}}{q' + 1p_{0}}\right) \qquad (1.25b)$$

 $n_{0} = p_{0} - \frac{4}{3} \wedge p_{0}^{3}$ (1.25c)

The solution given by (1.25) is the NUT solution generalized by the presence of the cosmological constant. The parameter n_0 coincides with p_0 when $\Lambda = 0$. This parameter is the NUT parameter or magnetic mass parameter.

It is clear that the generalized family of NUT solutions described in the canonical form by (1.19) and obtained by a contraction from Plebanski spacetime (1.1) represents the combined NUT-Reissner-Nordstrom solution with the cosmological constant, additionally generalized by the possible presence of magnetic monopole. The parameters e and g have interpretation of the electric and magnetic charges; M and n_o have interpretation of the mass and the NUT parameter.

If we put $p_0 = 0$ in (1.21) then we have $ds^2 = q'^2 (d\theta^2 + \theta^2 d\phi^2) + \frac{dq'^2}{Y_1} - Y_1 dt'^2$... (1.26a) where

where

$$Y_1 = -\frac{\Lambda}{3}q'^2 - \frac{2M}{q'} + \frac{e^2 + g^2}{q'^2}$$
 ... (1.26b)

Equation (1.26) is the Kasner-type spacetime, an anisotropic universe . Setting $\Lambda = e = g = 0$, equation (1.26) can be transformed to Kasner form .

Now if we set $p_0 = 0$ in (1.23) then we have

$$ds^{2} = q'^{2} (d\theta^{2} + \sinh^{2} \theta d\phi^{2}) + \frac{dq'^{2}}{Y_{1}} - Y_{1} dt'^{2} \qquad \dots \qquad (1.27a)$$

where

$$Y_{1} = -1 - \frac{\Lambda}{3}q'^{2} - \frac{2M}{q'} + \frac{e^{2} + g^{2}}{q'^{2}} \qquad \dots \qquad (1.27b)$$

The solution (1.27) presents Levi-Civita's type of generalization of the cosmological Reissner - Nordstrom solution with charges of both types . The Levi - Civita's metric is sometimes interpreted as the metric of a heavy tachyon .

(b) Now we consider " generalized anti-NUT solutions " given by (1.17) and restrict the parameter γ_1 to the discrete values

$$\gamma_1 = 1, 0, -1$$

 $\gamma_1 = 1$ Case I:

If we put $\alpha_1 = 1$, $\beta_1 = 0$ and introduce new coordinates $x'^{\mu} = (p', \phi, \theta, \tau')$ in place of $x'^{\mu} = (p', \sigma', q', \tau')$ defined by -: 22 :-

$$p' = p'$$
, $\sigma' = \phi$, $q' = \sinh \theta$, $\tau' = \tau'$

then (1.17) reduces to the form

$$ds^{2} = \frac{d p'^{2}}{x_{2}} + x_{2} (d\tau' + 2q_{0} \sinh \theta d\phi)^{2} + (q_{0}^{2} + p'^{2}) (d\theta^{2} - \cosh \theta d\phi^{2}) \qquad \dots \qquad (1.29a)$$

where

$$X_{2} = -1 - \frac{\Lambda}{3} (p'^{2} + 5 q_{0}^{2}) + 2 \operatorname{Re} \left(\frac{m_{0} + in}{q_{0} + ip'}\right) - \left|\frac{e + ig}{q_{0} + ip'}\right|^{2} \dots (1.29b)$$

$$m_{0} = q_{0} + \frac{4}{3} \wedge q_{0}^{3} \dots (1.29c)$$

Case II:
$$\gamma_1 = 0$$

If we now take $a_1 = 1$, $\beta_1 = 0$ and introduce new coordinates $x''^{\mu} = (p', y, x, \tau')$ for $x'^{\mu} = (p', \sigma', q', \tau')$ defined by

$$p' = p'$$
, $\sigma' = y$, $q' = x$, $\tau' = \tau'$... (1.30)

then (1.17) reduces to the form

$$ds^{2} = \frac{d p'^{2}}{x_{2}} + x_{2} (d\tau' + 2q_{0} x dy)^{2} + (q_{0}^{2} + p'^{2}) (dx^{2} - dy^{2}) \cdots (1.31a)$$

-: 23 :-

where

$$X_{2} = -\frac{\Lambda}{3}(p'^{2} + 5q_{0}^{2}) + 2 \operatorname{Re}\left(\frac{m_{0} + \operatorname{in}}{q_{0} + \operatorname{i}p'}\right) - \left|\frac{e + \operatorname{ig}}{q_{0} + \operatorname{i}p'}\right|^{2} \dots (1.31b)$$

$$m_{0} = \frac{4}{3} \wedge q_{0}^{3} \dots (1.31c)$$

Case III:
$$\gamma_1 = -1$$

In this case if we set $a_1 = -1$, $\beta_1 = 0$ and consider the transformation defined by

$$p' = p', \sigma' = \phi, q' = \sinh \theta, \tau' = \tau'$$
 ... (1.32)

then we have (1.17) as

$$ds^{2} = \frac{d p'^{2}}{X_{2}} + X_{2} (d_{\tau}' + 2q_{0} \sinh \theta d\phi)^{2}$$
$$- (q_{0}^{2} + p'^{2}) (d_{\theta}^{2} - \cosh^{2} \theta d\phi^{2}) \qquad \dots (1.33a)$$

where

$$X_{2} = 1 - \frac{\Lambda}{3} (p'^{2} + 5q_{o}^{2}) + 2 \operatorname{Re} \left(\frac{m_{o} + in}{q_{o} + ip'} - \left|\frac{e + ig}{q_{o} + ip'}\right|^{2} \dots (1.33b)$$

$$m_{o} = -q_{o} + \frac{4}{3} \wedge q_{o}^{3} \dots (1.33c)$$

Equations (1.29), (1.31) and (1.33) are the canonical representations of the anti-NUT solution.

-: 24 :-

If we put
$$e = g = q = \Lambda = 0$$
, then the equation (1.33) takes the form

$$ds^{2} = \frac{dp'^{2}}{x_{2}} + x_{2} d\tau'^{2} - p'^{2} (d\theta^{2} - \cosh^{2}\theta d\phi^{2}) \qquad \dots \qquad (1.34a)$$

where

$$X_2 = 1 + \frac{2n}{p'}$$
 ... (1.34b)

.

which is a solution of Einstein's equation in vacuum.

1.4 The Combined NUT - Kerr - Newman - Kasuya Spacetime

,

$$\varepsilon = 1$$
, $\Lambda = 0$, $b = a^2 - n^2 + g^2$... (1.35)

then the metric (1.1) reduces to the form

.

$$ds^{2} = \frac{p^{2} + q^{2}}{x} dp^{2} + \frac{x}{p^{2} + q^{2}} (d\tau + q^{2} d\sigma)^{2} + \frac{p^{2} + q^{2}}{y} dq^{2} - \frac{y}{p^{2} + q^{2}} (d\tau - p^{2} d\sigma)^{2} \qquad \dots (1.36a)$$

where

$$X = a^{2} - (n - p)^{2} \qquad (1.36b)$$

-: 25 :-

(1.36c)

$$Y = q^2 - 2Mq + a^2 - n^2 + e^2 + g^2$$

and the parameter a has the interpretation of angular momentum per unit mass. The above equation represents the combined NUT -Kerr - Newman - Kasuya spacetime in Boyer coordinates. For n = e = g = 0 the equation (1.36) reduces to the Kerr spacetime [29].

If we perform the coordinate transformation

then we obtain from (1.36)

$$ds^{2} = \Sigma d\theta^{2} + \frac{\Sigma}{Y} dr^{2} + \frac{\sin^{2} \theta}{\Sigma} (a dt - \rho d\phi)^{2}$$
$$- \frac{Y}{\Sigma} (dt - A d\phi)^{2} \qquad \dots (1.38a)$$

where

$$\Sigma = r^{2} + (n + a \cos \theta)^{2}$$

$$Y = r^{2} - 2Mr + a^{2} - n^{2} + e^{2} + g^{2}$$

$$\rho = r^{2} + a^{2} + n^{2}$$

$$A = a \sin^{2} \theta - 2n \cos \theta$$
.
(1.38b)

-: 26 :-

Equation (1.38) represents the NUT - Kerr - Newman - Kasuya (NUTKNK) spacetime in Boyer - Lindquist coordinates. The NUTKNK spacetime includes

(i) Kerr - Newman - Kasuya spacetime [30] when n = 0(ii) NUT - Kerr - Newman spacetime for g = 0[31] with e = g = 0(iii) NUT - Kerr spacetime (iv) Kerr - Newman spacetime [32] if $\mathbf{n} = \mathbf{g} = \mathbf{0}$ [33] provided n = g = e = 0(v) Kerr spacetime (vi) Reissner - Nordstrom spacetime [34,35] for n = g = a = 0(vii) Schwarzschild spacetime [36] when n = g = a = e = 0[37] if a = g = 0(viii) charged NUT spacetime a = e = g = 0. (ix) NUT spacetime [38] for

So we see that the NUTKNK spacetime includes all the black hole spacetimes (iv) - (vii), which are asymptotically flat. In particular the NUTKNK spacetime contains the NUT spacetime which has peculiar properties.

1.5 The Combined NUT - Kerr - Newman - Kasuya - de Sitter Spacetime

Setting

$$\varepsilon = 1 - \frac{\Lambda}{3} (a^2 + 6n^2)$$

$$b = a^2 - n^2 + g^2 - \frac{5}{3} \wedge n^2 (a^2 + n^2)$$

(1.39)

and replacing n by

$$n + \frac{\lambda}{3} \left(\frac{3 a^2 n^2}{p} + \frac{2 n^4}{p} + 2n^3 - 6n^2 p - a^2 n + 2n p^2 \right) \qquad \dots \qquad (1.40)$$

the metric (1.1) can be brought to the form

$$ds^{2} = \frac{p^{2} + q^{2}}{X} dp^{2} + \frac{X}{p^{2} + q^{2}} (d\tau + q^{2}d\sigma)^{2} + \frac{p^{2} + q^{2}}{Y} dq^{2}$$
$$- \frac{Y}{p^{2} + q^{2}} (d\tau - p^{2}d\sigma)^{2} \qquad \dots (1.41a)$$

where

•

$$X = \left[a^{2} - (n - p)^{2}\right] \left[1 + \frac{\lambda}{3}(n - p)^{2}\right] \qquad \dots \qquad (1.41b)$$

$$Y = (q^{2} + a^{2} + n^{2}) [1 - \frac{\Lambda}{3} (q^{2} + 5n^{2})] - 2 (Mq + n^{2}) + e^{2} + g^{2} \qquad \dots (1.41c)$$

Equation (1.41) represents the combined NUT - Kerr-Newman - Kasuyade Sitter spacetime [39] in Boyer coordinates. -: 28 :-

If we now perform the coordinate transformation

 $p = n + a \cos \theta , \quad \sigma = - \exists^{-1} (\phi / a) \qquad \dots \qquad (1.42)$ $q = r , \quad \tau = \exists^{-1} [t - \frac{(n^2 + a^2)}{a} \phi]$ where $\exists = 1 + \frac{\Lambda}{3} a^2$

then the metric (1.41) can be transformed to the form

$$ds^{2} = \frac{\Sigma}{\Delta_{\theta}} d\theta^{2} + \frac{\Sigma}{\Delta_{r}} dr^{2} + \frac{\frac{\pi^{2}}{2}}{\Sigma} (adt - \rho d\phi)^{2}$$

$$- \frac{\frac{\pi^{2}}{2}}{\Sigma} (dt - A d\phi)^{2} \dots (1.43a)$$

where

$$\Sigma = r^{2} + (n + a \cos \theta)^{2}$$

$$\Delta_{\theta} = 1 + \frac{\Lambda}{3} a^{2} \cos^{2} \theta \qquad (1.43b)$$

$$\Delta_{r} = (r^{2} + a^{2} + n^{2}) [1 - \frac{\Lambda}{3} (r^{2} + 5 n^{2})] - 2 (Mr + n^{2}) + e^{2} + g^{2}$$

 $\rho = r^2 + a^2 + n^2$

$$A = a \sin^2 \theta - 2n \cos \theta$$

The equation (1.43) represents the combined NUT - Kerr - Newman -Kasuya - de Sitter spacetime [40] in Boyer - Lindquist coordinates. We call the metric (1.43) as hot NUT - Kerr - Newman-Kasuya (HNUTKNK) spacetime since the de Sitter spacetime has been interpreted as being hot [41]. The HNUTKNK spacetime includes :

(i) NUTKNK spacetime for
$$\Lambda = 0$$

(ii) hot Kerr - Newman - Kasuya (HKNK) spacetime when n = 0(iii) hot NUT - Kerr - Newman (HNUTKN) spacetime for g = 0(iv) hot Kerr - Newman spacetime [42] with n = g = 0(v) hot Kerr spacetime [43] if n = g = e = 0(vi) hot Reissner - Nordstrom spacetime for n = g = a = 0(vii) hot Schwarzschild spacetime [44] for n = g = a = e = 0(viii) hot NUT spacetime [45] for a = e = g = 0.

So we see that the HNUTKNK spacetime includes the NUTKNK, HKNK, HNUTKN, hot NUT spacetimes as well as all the black hole spacetimes (iv) - (vii) which are asymptotically de Sitter. Further if we set $\wedge = 0$, in the cases (ii) - (vii) we get the Kerr-Newman - Kasuya, NUT - Kerr - Newman spacetimes and all the black hole spacetimes which are asymptotically flat. In the limit $\wedge = 0$, the case (viii) reduces to the NUT spacetime which is considered as homogeneous anisotropic cosmological model [46].

Thus the Plebanski spacetime (1.1) contains a large number of solutions of Einstein - Maxwell equations with or without cosmological constant which are important from the physical point of view. The metric (1.1) contains some spacetimes with cosmological parameter which may be found interesting from the point of view of its inflationary scenario of the early universe [47].

1.6 A Modified form of the Plebenski Spacetime

In this section we present a new form of the Plebanski spacetime. We show that a simple coordinate transformation brings the Plebanski spacetime to a form which includes an additional parameter other than the parameters present in the Plebanski spacetime. For this purpose if we introduce the new coordinate (p', o', q', τ') in place of the coordinate (p, o, q, τ) defined by

p' = p, q' = q, $\tau' = v\tau$, d' = vd ... (1.44)

where v is a real constant , then the metric (1.1) takes the form

-: 31 :-

$$ds^{2} = \frac{p^{2} + q^{2}}{x} dp^{2} + \frac{p^{2} + q^{2}}{y} dq^{2} + \frac{x}{\sqrt{2}(p^{2} + q^{2})} (d\tau + q^{2} d\sigma)^{2}$$

$$- \frac{Y}{\sqrt{2}(p^{2} + q^{2})} (d\tau - p^{2} d\sigma)^{2} \qquad \dots \qquad (1.45a)$$

$$X = b - g^{2} + 2np - \varepsilon p^{2} - \frac{\Lambda}{3} p^{4} \qquad \dots \qquad (1.45b)$$

$$Y = b + e^{2} - 2Mq + \varepsilon q^{2} - \frac{\Lambda}{3} q^{4} \qquad \dots \qquad (1.45c)$$

where we have dropped the primes .

The equation (1.45) can be identified as the modified form of the Plebanski spacetime [48] in Boyer coordinates. The parameters M, n, e, g have the same interpretations as in (1.1) and the parameter v is the adjustable parameter like the parameters b and ε in (1.1). By the similar arguments as given in [49] one can show that the metric (1.45) represents a class of solutions of Einstein - Maxwell equations.

For v = 1, the metric (1.45) gives the original Plebanski metric (1.1).

Setting

$$v = 1 + \frac{\Lambda}{3} a^{2}, \ \varepsilon = 1 - \frac{\Lambda}{3} (a^{2} + 6n^{2})$$

$$b = a^{2} - n^{2} + g^{2} - \frac{5}{3} \wedge n^{2} (a^{2} + n^{2})$$
(1.46)

-: 32 :-

and replacing n by

~

$$n + \frac{\Lambda}{3} \left(\frac{3 a^2 n^2}{p} + \frac{2 n^4}{p} + 2 n^3 - 6 n^2 p - a^2 n + 2 n p^2 \right) \dots (1.47)$$

and then accomplishing the coordinate transformation

$$p = n + a \cos \theta$$
, $\sigma = -\phi/a$
 $q = r$, $\tau = t - \frac{(n^2 + a^2)}{a}$, (1.48)

the metric (1.45) can be brought to the form

$$ds^{2} = \frac{\Sigma}{P} d\theta^{2} + \frac{\Sigma}{Y} dr^{2} + \frac{P \sin^{2} \theta}{(1 + \frac{\Lambda}{3}a^{2})^{2}\Sigma} (a dt - \rho d\phi)^{2}$$

$$-\frac{Y}{(1+\frac{\Lambda}{3}a^2)^{2}\Sigma} (dt - \Lambda d\phi)^2 \qquad ... (1.49a)$$

where

$$\Sigma = r^{2} + (n + a \cos \theta)^{2}$$

$$P = 1 + \frac{\Lambda}{3} a^{2} \cos^{2} \theta$$

$$Y = (r^{2} + a^{2} + n^{2}) [1 - \frac{\Lambda}{3} (r^{2} + 5n^{2})] - 2 (Mr + n^{2}) + e^{2} + g^{2}$$

$$\rho = r^{2} + a^{2} + n^{2}$$

$$A = a \sin^{2} \theta - 2n \cos \theta$$

-: 34 :-

CHAPTER - 2

SEPARATED DIRAC EQUATION IN AN ARBITRARY CURVED SPACETIME

The separability properties of Hamilton - Jacobi and Klein - Gordon equations have been studied by Carter [50,51] in the Kerr-Newman spacetime . Debever et al . [52] investigated the separability properties of the Hamilton - Jacobi and Klein - Gordon equations in an arbitrary curved background spacetime . The separability of Weyl neutrino equation, Maxwell's equations and the perturbed Einstein gravitational field equations has been established by Teukolsky [53] in the Kerr spacetime and using an analogous method , in the Plebanski - Demianski back -[54] by Dudley and Finley [55]. Teukolsky's separation ground of the variables of the equations governing the electromagnetic , the gravitational and the two-component neutrino - field perturbations of Kerr black hole has been central to much of the later developments . Following Teukolsky's separation method, Chandrasekhar [56] separated Dirac's equation in the Kerr spacetime and showed that the solutions can be expressed in terms of certain radial and angular function (satisfying decoupled equations). Chandrasekhar's result was immediately extended by Page [57] to the Kerr - Newman spacetime and subsequently analyzed by Carter and McLenaghan [58,59]. Recently Kamran and McLenaghan [60] performed a systematic study of the separability properties of neutrino and massive charged Dirac equations and obtained a system of coupled first order ordinary differential equations and a system of decoupled second order ordinary differential equations .

In this chapter we study the separability properties of the massive charged Dirac equation in an arbitrary curved background spacetime [61] and obtain the separated Dirac equation as a system of decoupled second order ordinary differential equation from which the pertinent equation will be used to derive the radial equation for the metric concerned to study the Hawking radiation in the later chapters.

Finally, we show how the separated Dirac equation of Page [62] in the Kerr-Newman background and of Chandrasekhar [63] in the Kerr background emerge from the general expression of the separated Dirac equation in an arbitrary curved background spacetime as special cases.

2.1 Arbitrary Curved Spacetime and Some Preliminaries

The general structures for the spacetimes which are solutions of Einstein-Maxwell equations with cosmological constant have been established by Debever and McLenaghan [64,65] and Diaz and Salazar I [66]. But recently Debever el al. [67,68] and Diaz [69] determined, in a systematic and unified manner, the complete set of solutions of Petrov type D to the Einstein - Maxwell equations with cosmological constant. The obtained solutions contain, all known type D metrics: for instance Plebanski - Demianski solution [70], Plebanski solution [71], Kinnersley solution [72] Carter solutions [73,74,75] and so forth. -: 36 :-

For our purpose we consider the spacetime [76] in the coordinates (u, v, w, x)

$$ds^{2} = \frac{Z}{T^{2}} [fw^{2} z^{2} (\epsilon_{1} du + m dv)^{2} + (1 - f^{2})g^{-2} z^{-1} (\epsilon_{1} du + m dv) dw$$

-f g^{-4} w^{-2} dw^{2} - x^{2} z^{-2} (\epsilon_{2} du + p dv)^{2} - x^{-2} dx^{2}] ... (2.1a)

with

$$A = A_{\mu} dx^{\mu} = \frac{1}{Z} \left[\left(\epsilon_2 G + \epsilon_1 H \right) du + \left(p G + m H \right) dv \right] \qquad \dots \qquad (2.1b)$$

where

$$W = W(w) , X = X(x) , T = T(w, x) , G = G(x) , H = H(w) ,$$

$$p = p(w) , m = m(x) , g = [(1 + f^{2}) / 2]^{1/2} ... (2.1c)$$

$$Z = Z(w, x) = \varepsilon_{1} p(w) - \varepsilon_{2} m(x)$$
and $\varepsilon_{1} , \varepsilon_{2}$ and f are real constants satisfying $\varepsilon_{1}^{2} + \varepsilon_{2}^{2} \neq 0$
and all functions are real valued.

By specializing the constants and the functions appearing in (2,1), a particular class of spacetimes of the metric may be deduced. For example if we set in equation (2.1) the following substitutions :

 $\varepsilon_1 = \varepsilon_2 = f = g = T = 1$, $Z = w^2 + x^2$, $p = w^2$, $m = -x^2$, $X = \sqrt{X}$, $W = \sqrt{Y}$ where X and Y are given by (1.1b) and (1.1c), we get the Plebanski spacetime. -: 37 :-

The spin coefficients for the metric (2.1) are given by [77]

$$\rho = (\sqrt{2})^{-1} W T z^{-1/2} [T^{-1} T_{W} - \frac{1}{2} z^{-1} z_{W} + \frac{1}{2} z^{-1} \varepsilon_{1} m_{X}]$$

$$\tau = (\sqrt{2})^{-1} X T z^{-1/2} [\frac{1}{2} z^{-1} \varepsilon_{2} p_{W} + i(\frac{1}{2} z^{-1} z_{X} - T^{-1} T_{X})]$$

$$\varepsilon = \frac{1}{2\sqrt{2}} W T z^{-1/2} [W^{-1} W' - \frac{1}{2} z^{-1} z_{W} - T^{-1} T_{W} + \frac{1}{2} z^{-1} \varepsilon_{1} m_{X}] (2.2)$$

$$\alpha = \frac{1}{2\sqrt{2}} X T z^{-1/2} [\frac{1}{2} z^{-1} \varepsilon_{2} p_{W} + i(T^{-1} T_{X} + \frac{1}{2} z^{-1} z_{X} - X^{-1} X)]$$

$$\mu = f\rho, \quad \gamma = f\varepsilon, \quad \pi = \tau, \quad \beta = \alpha, \quad K = \delta = \lambda = \nu = 0, \quad f^{2} = 1.$$

The derivative operators are given by

$$D = \frac{1}{\sqrt{2}} T z^{-1/2} \left[f g^{-2} W^{-1} \left(p \frac{\partial}{\partial u} - \varepsilon_2 \frac{\partial}{\partial v} \right) + W \frac{\partial}{\partial w} \right]$$

$$\Delta = \frac{1}{\sqrt{2}} \operatorname{T} z^{-1/2} \left[g^{-2} W^{-1} \left(p \frac{\partial}{\partial u} - \varepsilon_2 \frac{\partial}{\partial v} \right) - f W \frac{\partial}{\partial w} \right]$$
(2.3)

$$\delta = \frac{1}{\sqrt{2}} \operatorname{T} z^{-1/2} \left[x^{-1} \left(\varepsilon_1 \frac{\partial}{\partial u} - m \frac{\partial}{\partial v} \right) - i x \frac{\partial}{\partial x} \right]$$

-: 38 :-

2.2 Separation of the Dirac Equation

In the Newman - Penrose formalism the Dirac equation can be written in the form , using the Weyl representation for ψ

$$H_{D} \Psi = \begin{bmatrix} -i\mu_{\Theta} & 0 & D + \overline{\epsilon} - \overline{\rho} - iQA_{1} \\ 0 & -i\mu_{\Theta} & \overline{\delta} + \overline{\beta} - \overline{\tau} - iQA_{4} \\ \Delta + \mu - \gamma - iQA_{2} & -(\delta + \beta - \tau) + iQA_{3} & -i\mu_{\Theta} \\ -(\overline{\delta} + \pi - \alpha) + iQA_{4} & D + \epsilon - \rho - iQA_{1} & 0 \end{bmatrix}$$

$$\begin{split} \mathbf{b} + \mathbf{\bar{n}} - \mathbf{\bar{\alpha}} - \mathbf{i}\mathbf{Q}\mathbf{A}_{3} \\ \mathbf{\Delta} + \mathbf{\bar{\mu}} - \mathbf{\bar{\gamma}} - \mathbf{i}\mathbf{Q}\mathbf{A}_{2} \\ \mathbf{0} \\ - \mathbf{i}\mu_{e} \end{split} \begin{bmatrix} \mathbf{P}_{\mathbf{0}} \\ \mathbf{P}_{\mathbf{1}} \\ \mathbf{\bar{q}}^{\mathbf{0}} \\ \mathbf{\bar{q}}^{\mathbf{1}} \end{bmatrix} = \begin{bmatrix} \mathbf{0} \\ \mathbf{0} \\ \mathbf{0} \\ \mathbf{\bar{q}}^{\mathbf{1}} \end{bmatrix}$$

(2.4)

where μ_{Θ} and Q are the mass and the electric charge of the Dirac particle respectively .

We have from (2.1b), (2.2) and (2.3)

$$D + \varepsilon - p - iQA_{1} = \frac{1}{\sqrt{2}} T z^{-1/2} \left[W \frac{\partial}{\partial w} + f g^{-2} \left(p \frac{\partial}{\partial u} - \varepsilon_{2} \frac{\partial}{\partial v} \right) \right]$$

$$- \frac{1}{4} W z^{-1} \varepsilon_{1} \left(-p' + im' \right) + \frac{1}{2} W' - \frac{3}{2} W T^{-1} T_{W} \right]$$

$$- iQ z^{-1} \left(\varepsilon_{2} Q + \varepsilon_{1} H \right) \qquad (2.5a)$$

$$\Delta + \mu - \gamma - iQA_{2} = \frac{1}{\sqrt{2}} T z^{-1/2} \left[-fW \frac{\partial}{\partial w} + g^{-2} W^{-1} \left(p \frac{\partial}{\partial u} - \frac{\varepsilon}{2} \frac{\partial}{\partial v} \right) \right]$$

$$+ \frac{1}{4} f W z^{-1} \varepsilon_{1} \left(-p' + im' \right) - \frac{1}{2} f W'$$

$$+ \frac{3}{2} f W T^{-1} T_{W} \right] - iQ z^{-1} \left(p Q + m H \right) \qquad (2.5b)$$

$$b + \beta - \tau - iQA_{3} = \frac{1}{\sqrt{2}} T z^{-1/2} \left[-iX \frac{\partial}{\partial x} + x^{-1} \left(\varepsilon_{1} \frac{\partial}{\partial u} - m \frac{\partial}{\partial v} \right) \right]$$

$$- \frac{1}{4} X z^{-1} \varepsilon_{2} \left(p' - im' \right) - \frac{1}{2} X' + \frac{3}{2} iX T^{-1} T_{X} \right] (2.5c)$$

Let us perform the transformation

$$\Psi = S \Psi' \qquad (2.6a)$$

where

.

$$S = T^{3/2} Z^{-1/4} \operatorname{diag} (e^{iB}, e^{iB}, e^{-iB}, e^{-iB}) \dots (2.6b)$$
$$dB = (4 Z)^{-1} (\epsilon_1 m' dw + \epsilon_2 p' dx) \dots (2.6c)$$

and separating matrix defined by

$$U = Z^{1/2} T^{-1} \operatorname{diag} (e^{2iB}, -e^{2iB}, -e^{-2iB}, e^{-2iB}) \quad (2.7)$$

We obtain from (2.4), (2.6b) and (2.7)

$$W_{D} = U S^{-1} H_{D} S$$

$$= \begin{bmatrix} -i\mu_{e} Z^{1/2} T^{-1} e^{2iB} & 0 & D_{w}^{+} & D_{x}^{-} \\ 0 & i\mu_{e} Z^{1/2} T^{-1} e^{2iB} & -D_{x}^{-} & -D_{w}^{-} \\ - D_{w}^{-} & D_{x}^{-} & i\mu_{e} Z^{1/2} T^{-1} e^{-2iB} & 0 \\ - D_{w}^{+} & D_{w}^{+} & 0 & -i\mu_{e} Z^{1/2} T^{-1} e^{-2iB} \end{bmatrix}$$

(2.8)

-: 41 :-

where the operators

$$D_{x}^{+} = \frac{1}{\sqrt{2}} \left[1 \left(x \frac{\partial}{\partial x} + \frac{x'}{2} - \frac{QG}{x} \right) + \frac{1}{x} \left(\varepsilon_{1} \frac{\partial}{\partial v} - m \frac{\partial}{\partial u} \right) \right] \dots (2.9a)$$

$$\mathbf{D}_{\mathbf{X}} = \frac{1}{\sqrt{2}} \left[-\mathbf{i} \left(\mathbf{X} \frac{\partial}{\partial \mathbf{x}} + \frac{\mathbf{X}'}{2} + \frac{\mathbf{QG}}{\mathbf{X}} \right) + \frac{1}{\mathbf{X}} \left(\varepsilon_{1} \frac{\partial}{\partial \mathbf{v}} - \mathbf{m} \frac{\partial}{\partial \mathbf{u}} \right) \right] \dots (2.9b)$$

$$D_{W}^{+} = \frac{1}{\sqrt{2}} \left[W \frac{\partial}{\partial W} + \frac{W'}{2} - \frac{i f Q H}{g^{2} W} + \frac{f}{g^{2} W} \frac{\partial}{\partial u} - \frac{\partial}{2 \partial v} \right] \dots (2.9c)$$

$$D_{W}^{-} = \frac{1}{\sqrt{2}} \left[-f_{W} \frac{\partial}{\partial w} - \frac{f_{W}}{2} - \frac{i_{Q}H}{g^{2}W} + \frac{1}{g^{2}W} \left(p \frac{\partial}{\partial u} - \varepsilon_{2} \frac{\partial}{\partial v} \right) \right] \dots (2.9d)$$

The separability condition for the transformed equation $W_D \psi' = 0$ is that

$$z^{1/2} T^{-1} e^{2iB} = [f(x) + ig(x)] + [h(w) + ik(w)]$$
 (2.10)

where f(x), g(x), h(w) and k(w) are real valued functions. But this condition is not sufficient for separability with ψ' given by

$$\psi^{\prime}(u,v,w,x) = e^{i(\alpha u + \beta v)} \begin{bmatrix} H_{1}(x) & K_{2}(w) \\ H_{2}(x) & K_{1}(w) \\ H_{1}(x) & K_{1}(w) \\ H_{2}(x) & K_{2}(w) \end{bmatrix}$$
(2.11)

where α and β are arbitrary constants.

Substituting (2.10) and (2.11) into $W_D \psi' = 0$ with W_D as given by (2.8) we obtain the following equations :

$$\overset{O^{+}}{D}_{w} \overset{W}{l} (w) \overset{H}{l} (x) - i\mu_{\Theta} [h(w) + ik(w)] \overset{K}{K}_{2} (w) \overset{H}{l} (x)$$

$$+ \overset{O^{-}}{D}_{x} \overset{H}{l}_{2} (x) \overset{K}{k}_{2} (w) - i\mu_{\Theta} [f(x) + ig(x)] \overset{H}{l} (x) \overset{K}{k}_{2} (w) = 0 \quad (2.12a)$$

$$D_{w}^{O} K_{2}(w) H_{2}(x) - i \mu_{\theta} [h(w) + i k(w)] K_{1}(w) H_{2}(x) + D_{x}^{O+} H_{1}(x) K_{1}(w) - i \mu_{\theta} [f(x) + i g(x)] H_{2}(x) K_{1}(w) = 0 \quad (2.12b)$$

$$\overset{O}{D}_{w}^{-} K_{2}(w) H_{1}(x) - i \mu_{e} [h(w) - i k(w)] K_{1}(w) H_{1}(x)$$

$$- \overset{O}{D}_{x}^{-} H_{2}(x) K_{1}(w) - i \mu_{e} [f(x) - i g(x)] H_{1}(x) K_{1}(w) = 0 \quad (2.12c)$$

$$D_{w}^{0+} K_{1}(w) H_{2}(x) - i \mu_{\theta} [h(w) - i k(w)] K_{2}(w) H_{2}(x)$$

- $D_{x}^{0+} H_{1}(x) K_{2}(w) - i \mu_{\theta} [f(x) - i g(x)] H_{2}(x) K_{2}(w) = 0$ (2.12d)

where the operators

. •

.

$$D_{x}^{0+} = \frac{1}{\sqrt{2}} \left[X \frac{\partial}{\partial x} + \frac{X'}{2} - \frac{QG}{X} + \frac{1}{X} (\beta \epsilon_{1} - \alpha m) \right] \qquad (2.12e)$$

$$D_{x}^{0} = \frac{-1}{\sqrt{2}} \left[X \frac{\partial}{\partial x} + \frac{X'}{2} + \frac{QG}{X} - \frac{1}{X} (\beta \epsilon_{1} - \alpha m) \right] \qquad (2.12f)$$

$$D_{W}^{0+} = \frac{1}{\sqrt{2}} \left[W \frac{\partial}{\partial W} + \frac{W'}{2} - \frac{if QH}{g^{2}W} + \frac{if}{g^{2}W} (\alpha p - \beta \varepsilon_{2}) \right] \qquad (2.12g)$$

$$D_{W}^{\circ} = \frac{1}{\sqrt{2}} \left[-f_{W} \frac{\partial}{\partial w} - \frac{f_{W}}{2} - \frac{i_{QH}}{g^{2}_{W}} + \frac{i}{g^{2}_{W}} - \frac{i_{QH}}{g^{2}_{W}} - \frac{i_{QH}}{g^{2}_{W}} + \frac{i}{g^{2}_{W}} - \frac{i_{QH}}{g^{2}_{W}} - \frac{i_{QH}}{g^{2}_{$$

.

The relations
$$(2.12a) - (2.12d)$$
 give then immediately
 $\overset{0}{P}_{w}^{+} K_{1}(w) - i \mu_{e} [h(w) + ik(w)] K_{2}(w) = \mu_{1}K_{2}(w) \dots (2.13a)$
 $\overset{0}{P}_{x}^{-} H_{2}(x) - i \mu_{e} [f(x) + i g(x)] H_{1}(x) = -\mu_{1}H_{1}(x) \dots (2.13b)$
 $\overset{0}{P}_{w}^{-} K_{2}(w) - i \mu_{e} [h(w) + i k(w)] K_{1}(w) = \mu_{2}K_{1}(w) \dots (2.13c)$
 $\overset{0}{P}_{x}^{+} H_{1}(x) - i \mu_{e} [f(x) + i g(x)] H_{2}(x) = -\mu_{2}H_{2}(x) \dots (2.13c)$
 $\overset{0}{P}_{w}^{-} K_{2}(w) - i \mu_{e} [h(w) - i k(w)] K_{1}(w) = \mu_{3}K_{1}(w) \dots (2.13c)$
 $\overset{0}{P}_{w}^{-} H_{2}(x) - i \mu_{e} [f(x) - i g(x)] H_{1}(x) = -\mu_{3}H_{1}(x) \dots (2.13c)$
 $\overset{0}{P}_{w}^{+} K_{1}(w) - i \mu_{e} [h(w) - i k(w)] K_{2}(w) = \mu_{4}K_{2}(w) \dots (2.13c)$
 $\overset{0}{P}_{w}^{+} K_{1}(w) - i \mu_{e} [f(x) - i g(x)] H_{2}(x) = -\mu_{4}H_{2}(x) \dots (2.13c)$

Subtraction of (2.13g) from (2.13a) yields

$$k(w) = C_1$$
 ... (2.14a)

and the addition of (2.13h) to (2.13d) yields

$$f(x) = C_2$$
 ... (2.14b)

where C_1 and C_2 are real constants.

If we now define $g_1(x)$ and $h_1(w)$ by

$$g_{1}(x) = g(x) + C_{1}$$
 ... (2.15a)

and

$$h_1(w) = h(w) + C_2$$
 ... (2.15b)

then the condition (2.10) can be reduced, after dropping the suffixes, to the form

$$Z^{1/2} - \frac{1}{2} e^{2iB} = h(w) + ig(x) \qquad \dots \qquad (2.16)$$

which is the separability condition for the existence of a separable solution .

Substituting (2.16) into $W_D \psi' = 0$ with W_D as given by (2.8) we obtain

$$\begin{bmatrix} -i\mu_{\theta} [h(w) + ig(x)] & 0 & p_{w}^{+} \\ 0 & i\mu_{\theta} [h(w) + ig(x)] & -p_{x}^{-} \\ -p_{w}^{-} & p_{x}^{-} & i\mu_{\theta} [h(w) + ig(x)] \\ -p_{x}^{0} & p_{w}^{+} & 0 \\ & p_{x}^{0} & p_{w}^{0} & 0 \\ & p_{x}^{0} & p_{w}^{0} & p_{w}^{0} \\ & p_{x}^{0} & p_{w}^{0} & p_{w$$

or,
$$\begin{bmatrix} -i\mu_{\theta}h(w) + \mu_{\theta}g(x) & 0 & D_{w}^{0+} \\ 0 & i\mu_{\theta}h(w) - \mu_{\theta}g(x) & -D_{x}^{0+} \\ & 0 & D_{w}^{0-} & D_{x}^{0-} \\ & -D_{w}^{0-} & D_{x}^{0-} & i\mu_{\theta}h(w) - \mu_{\theta}g(x) \\ & -D_{w}^{0+} & D_{w}^{0+} & 0 \end{bmatrix}$$

$$\begin{bmatrix} D_{x}^{0} \\ -D_{w}^{0} \\ 0 \\ -D_{w}^{0} \\ 0 \\ -D_{w}^{0} \\ -D_{w}^{0} \\ 0 \\ -D_{w}^{0} \\ -D_{w}^{0} \\ 0 \\ -D_{w}^{0} \\ -D_{w}^{0}$$

from which we obtain

$$\frac{1}{K_2(w)} \stackrel{0_+}{\longrightarrow} K_1(w) + \frac{1}{H_1(x)} \stackrel{0_-}{\longrightarrow} H_2(x) = i \mu_e h(w) + \mu_e g(x)... \quad (2.17a)$$

$$\frac{1}{H_{2}(x)} \stackrel{o_{+}}{\xrightarrow{D}_{x}} H_{1}(x) + \frac{1}{K_{1}(w)} \stackrel{o_{-}}{\xrightarrow{D}_{w}} K_{2}(w) = i\mu_{e}h(w) - \mu_{e}g(x) \dots (2.17b)$$

$$\frac{1}{K_{1}(w)} \bigcup_{w}^{O} K_{2}(w) - \frac{1}{H_{1}(x)} \bigcup_{x}^{O} K_{2}(x) = i\mu_{e}h(w) - \mu_{e}g(x) \dots (2.17c)$$

$$\frac{1}{H_{2}(x)} \stackrel{o_{+}}{\overset{b_{+}}{x}} H_{1}(x) - \frac{1}{K_{2}(w)} \stackrel{o_{+}}{\overset{b_{+}}{w}} K_{1}(w) = -i \mu_{\theta} h(w) - \mu_{\theta} g(x)$$
(2.17d)

where λ_1 , λ_2 , λ_3 and λ_4 are the separation constants. However, it is manifest that the consistency of the foregoing equations requires that

$$\lambda_1 = \lambda_2 = -\lambda_3 = \lambda_4 = \lambda$$
 (2.19)

We are thus left with the following equations :

$$D_{w}^{0+} K_{1}(w) - i \mu_{0} h(w) K_{2}(w) = \lambda K_{2}(w) \qquad (2.20 a)$$

$$D_{w}^{0-} K_{2}(w) - i \mu_{0} h(w) K_{1}(w) = -\lambda K_{1}(w)$$

$$D_{x}^{0-} H_{2}(x) + \mu_{0} g(x) H_{1}(x) = -\lambda H_{1}(x)$$

$$D_{x}^{0+} H_{1}(x) + \mu_{0} g(x) H_{2}(x) = \lambda H_{2}(x)$$

$$(2.20 b)$$

The system of coupled first - order ordinary differential equations (2.20) can be written as a system of decoupled second-order ordinary differential equations in the following forms :

$$\sum_{W}^{O} \sum_{W}^{O+} K_{1}(w) - \frac{i \mu_{\Theta} \left[\sum_{W}^{O-} h(w) \right]}{\lambda + i \mu_{\Theta} h(w)} \sum_{W}^{O+} K_{1}(w)$$

$$+ \left[\lambda^{2} + \left\{ \mu_{\Theta} h(w) \right\}^{2} \right] K_{1}(w) = 0 \quad ... \quad (2.21a)$$

$$\overset{O^{+}}{D_{W}} \overset{O^{-}}{D_{W}} K_{2}(w) - \frac{i \mu_{\theta} [\overset{O^{+}}{D_{W}} h(w)]}{-\lambda + i \mu_{\theta} h(w)} \overset{O^{-}}{D_{W}} K_{2}(w)$$

$$+ [\lambda^{2} + \{\mu_{\theta} h(w)\}^{2}] K_{2}(w) = 0 \quad ... \quad (2.21b)$$

$$\sum_{x}^{o} \sum_{x}^{o} H_{1}(x) - \frac{\mu_{e} \left[\sum_{x}^{o} g(x) \right]}{-\lambda + \mu_{e} g(x)} \sum_{x}^{o} H_{1}(x)$$

$$- \left[\lambda^{2} - \left\{ \mu_{e} g(x) \right\}^{2} \right] H_{1}(x) = 0 \quad ... \quad (2.21c)$$

$$D_{x}^{o_{+}} D_{x}^{-} H_{2}(x) - \frac{\mu_{e} [D_{x}' g(x)]}{\lambda + \mu_{e} g(x)} D_{x}^{o_{-}} H_{2}(x)$$

+
$$[\lambda^2 - {\mu_0 g(x)}^2] H_2(x) = 0$$
 . . . (2.21d)

The equations(2.20) and (2.21) reduce to the separated equations of Chandrasekhar [78], Page [79], Guven [80] when the constants and the functions appearing in (2.1) are appropriately specialized.

Special Cases Reduction to 2.3

Let us now discuss some of the special cases of equations and (2.21). (2.20)

For the Kerr metric we obtain from equation (2.1), $\epsilon_1 = \epsilon_2 = f = g = 1, m(x) = -x^2 p(w) = w^2,$ T(w, x) = 1, H(w) = G(x) = 0, h(w) = w, $Z = w^{2} + x^{2}$, g(x) = x, $W(w) = \sqrt{Y}$, $X(x) = \sqrt{X}$, $X = a^2 - x^2$, $Y = w^2 - 2Mw + a^2$, $\alpha = id$, $\beta = -i(am_0 + da^2)$

M is the mass parameter , a is the angular momentum where per unit mass parameter , o is the energy of the Dirac particle and mo is any arbitrary constant.

From equations (2.12g,h), (2.20a) and (2.22) we obtain

$$(\sqrt{2})^{-1} \sqrt{Y} \left\{ \frac{\partial}{\partial w} + \frac{i K}{Y} - \frac{w - M}{2Y} \right\} K_{1}(w) = (\lambda + i \mu_{0} w) K_{2}(w)$$

$$(\sqrt{2})^{-1} \sqrt{Y} \left\{ \frac{\partial}{\partial w} - \frac{i K}{Y} + \frac{w - M}{2Y} \right\} K_{2}(w) = (\lambda - i \mu_{0} w) K_{1}(w)$$
where
$$Y = w^{2} - 2 Mw + a^{2},$$

$$K = (w^{2} + a^{2}) d + a m_{0}.$$

(2.22)

-: 49 :-

If we now allow the coordinate w to play the role of the new radial variable r, then replacing w by r and letting

$$K_1(r) = R_2 Y^{-1/4}$$
, $K_2(r) = (\sqrt{2})^{-1} R_1 Y^{1/4}$

we obtain (2.23), after simplification, as

$$\left(\begin{array}{c} \partial\\ --- + \mathbf{i} \mathbf{K}\end{array}\right) \mathbf{R}_{2} = \left(\lambda + \mathbf{i} \mu_{e} \mathbf{r}\right) \mathbf{R}_{1}$$

$$Y\left(\frac{\partial}{\partial r} - \frac{iK}{Y} + \frac{r-M}{Y}\right)R_{1} = 2\left(\lambda - i\mu_{e}r\right)R_{2}$$

where

$$Y = r^2 - 2Mr + a^2$$
,
 $K = (r^2 + a^2) \circ + am_0$,

The equations (2.24) correspond to Chandrasekhar's equations (40) [81] in the context of Kerr geometry.

Now from equations (2.12e,f), (2.20b) and (2.22) we obtain $(\sqrt{2})^{-1}\sqrt{X} \left[\frac{\partial}{\partial x} + \frac{am_{o} + \sigma (a^{2} - x^{2})}{X} - \frac{x}{2X}\right] H_{2}(x) = (\lambda + \mu_{e} x)H_{1}(x)$ (2.25) $(\sqrt{2})^{-1}\sqrt{X} \left[\frac{\partial}{\partial x} - \frac{am_{o} + \sigma (a^{2} - x^{2})}{X} - \frac{x}{2X}\right] H_{1}(x) = (\lambda - \mu_{e} x) (-i H_{2}(x))$

(2.24)

-: 50 :-

If
$$-i H_2(x) = S_1(\theta), H_1(x) = S_2(\theta)$$

and $x = a \cos \theta$ so that $X = a^2 \sin^2 \theta$

we obtain (2.25) as then

$$\left(\begin{array}{c} \frac{\partial}{\partial \theta} \\ \frac{\partial}{\partial \theta} \end{array}\right) = \sqrt{2} \left(\lambda + a\mu_{\theta} \cos\theta\right) S_{2}(\theta)$$

(2.26)

$$\begin{pmatrix} \frac{\partial}{\partial \theta} + m_{0} \cos \theta + a \theta \sin \theta + \frac{1}{-\cot \theta} S_{2}(\theta) = -\sqrt{2} (\lambda - a \mu_{0} \cos \theta) S_{1}(\theta) \\ \frac{\partial}{\partial \theta} + \frac{1}{2} \begin{pmatrix} \frac{\partial}{\partial \theta} + \frac{1}{2} + \frac{1}{2} & \frac{1}{2} \end{pmatrix}$$

The above equations correspond to those first derived by Chandrasekhar [82] in the context of Kerr geometry.

For the Kerr - Newman metric we set

$$\epsilon_{1} = \epsilon_{2} = f = g = 1, \quad m(x) = -x^{2}, \quad p(w) = w^{2},$$

$$T(w, x) = 1, \quad H(w) = w, \quad G(x) = 0, \quad h(w) = w,$$

$$Z = w^{2} + x^{2}, \quad g(x) = x, \quad W(w) = \sqrt{Y}, \quad X(x) = \sqrt{X}, \quad ... \quad (2.27)$$

$$X = a^{2} - x^{2}, \quad Y = w^{2} - 2 \quad Mw + a^{2} + e^{2}, \quad a = id,$$

$$\beta = -i(am_{o} + da^{2})$$

where M is the mass parameter , a is the angular momentum per unit mass parameter , e is the electric charge parameter, d is the energy of the Dirac particle and mois any constant. Then from equations (2.12g,h)(2.21a) and (2.27)we obtain the following equation :

$$\sqrt{Y} \left[-\frac{\partial}{\partial w} + \frac{i}{Y} - \frac{Y'}{4Y} \right] \sqrt{Y} \left[\frac{\partial}{\partial w} + \frac{i}{Y} + \frac{Y'}{4Y} \right] K_{1}(w) + \frac{i}{\lambda + i} \frac{\mu_{0}}{\mu_{0} w} - \left[-\frac{\partial}{\partial w} + \frac{i}{Y} + \frac{Y'}{4Y} \right] K_{1}(w) + 2 \left(\lambda^{2} + \mu_{0}^{2} w^{2} \right) K_{1}(w) = 0 \qquad (2.28)$$

where

$$Y = w^2 - 2 Mw + a^2 + e^2$$

 $K = (w^2 + a^2) \circ - e Qw + a m_0$

and the prime denotes differentiation with respect to the argument .

Now if w plays the role of the radial variable r, then replacing w by r and letting $K_1(r) = R Y^{-1/4}$, we find that the equation (2.28) reduces to an exceedingly compact form

$$\sqrt{Y} \quad \frac{d}{dr} \left(\sqrt{Y} \frac{dR}{dr} \right) - \frac{i \mu_{\theta} Y}{\lambda + i \mu_{\theta} r} \frac{dR}{dr} + \begin{bmatrix} \frac{K^2 - i(r - M)K}{r} \\ \frac{K^2 - i(r - M)K}{r} \end{bmatrix}$$

+ 21 or +
$$\frac{\mu_{\Theta}K}{\lambda + i\mu_{\Theta}r} - 2(\lambda^2 + \mu_{\Theta}^2r^2)]R = 0$$
 (2.29)

where

$$Y = r^{2} - 2Mr + a^{2} + e^{2},$$

K = (r² + a²) o - eQr + am₂.

If $\sqrt{2} \lambda \rightarrow \lambda$, $\sqrt{2} \mu_e \rightarrow \mu_e$, then the equation (2.29) corresponds to Page's equation (20) [83] in the context of Kerr - Newman spacetime.

For e = 0, the equation (2.29) reduces to the Chandrasekhar's equation (45) [84] in the case of Kerr metric.

When both e=0 and $\mu_{e}=0$, the equation (2.29) corresponds to Teukolsky's equation (4.10) [85] for neutrinos.

CHAPTER - 3

HAWKING RADIATION OF DIRAC PARTICLES IN NUT-KERR-NEWMAN SPACETIME

-: 53 :-

Hawking's [86] thermal radiation by black holes near black hole event horizon have been studied by different authors in different types of asymptotically flat background spacetimes . Chandrasekhar's [87] work provided the possibility of attacking the problem of Hawking evaporation of Dirac particles to the Kerr black hole spacetime . Liu Liao and [88] based on Chandrasekhar's work studied the Hawking Xu Dianyan evaporation of Dirac particles in the Kerr background . Page [89] extended Chandrasekhar's work to the Kerr - Newman black hole . Zhao Zheng et al. [90] based on Page's work extended the work of Liu Liao and Xu Dianyan to the spacetime describing the Kerr - Newman black hole. Ahmed [91] taking the advantage of the separability of the Dirac equation in an arbitrary curved background spacetime performed by Kamran and McLenaghan [92] extended the work of Zhao Zheng et al. to the NUT - Kerr - Newman (NUTKN) spacetime which includes all the black hole spactimes which are asymtotically flat . Ahmed's work are interesting in that Hawking's thermal radiation by black holes can also be obtained in the case of NUT spacetime which has peculiar properties . In this chapter we review Ahmed's work which serves a useful purpose in providing with a firm basis for the work on Hawking's thermal radiation near the event horizon of asymptotically flat background spacetimes . We use equation (2.21a) and derive the radial decoupled Dirac equation in the NUTKN limit and study Hawking radiation near the event horizon of NUTKN spacetime.

-: 24 :-

3.1 The NUTKN Spacetime

In the coordinates (u, v, w, x) the NUTKN spacetime is

$$ds^{2} = \frac{x^{2} + w^{2}}{x} dx^{2} + \frac{x^{2} + w^{2}}{y} dw^{2} + \frac{X}{x^{2} + w^{2}} (du + w^{2} dv)^{2}$$

$$- \frac{Y}{x^{2} + w^{2}} (du - x^{2} dv)^{2} \qquad (3.1a)$$

where

$$X = a^{2} - (n - x)^{2} \qquad (3.1b)$$

$$Y = w^{2} - 2Mw + a^{2} - n^{2} + e^{2} \qquad (3.1c)$$

with electric potential

$$A_{\mu} d Z^{\mu} = \frac{ew}{x^2 + w^2} (du - x^2 dv) \qquad (3.2)$$

The NUTKN spacetime contains four real parameters : the mass parameter M, the NUT (magnetic mass) parameter n , the angular momentum per unit mass parameter a , the electric charge parameter e. The NUTKN spacetime includes

(i) Kerr - Newman spacetime when n = 0

(ii) Kerr spacetime with n = e = 0

-: 55 :-

(iii) Reissner - Nordstrom spacetime if n = a = 0

(iv) Schwarzschild spacetime for n = a = e = 0

(v) NUT Kerr spacetime provided e = 0

(vi) NUT spacetime when a = e = 0.

Thus we see that the NUTKN spacetime contains all the black hole spacetimes (Kerr - Newman, Kerr, Reissner - Nordstrom, Schwarzschild) which are asymptotically flat.

3.2 Radial Wave Equation

To obtain the NUTKN spacetime and the radial decoupled Dirac equation in this background spacetime it suffices to set in equations (2.1), (2.21a), (2.12g) and (2.12h) the following substitutions :

$$\epsilon_{1} = \epsilon_{2} = f = g = 1, \quad m(x) = -x^{2}, \quad p(w) = w^{2},$$

$$T(w, x) = 1, \quad H(w) = ew, \quad G(x) = 0 \quad h(w) = w,$$

$$g(x) = x, \quad Z = w^{2} + x^{2}, \quad a = i\sigma, \quad \beta = -iA, \quad ... \quad (3.3)$$

$$A = a m_{o} + \sigma (a + n)^{2}, \quad W(w) = \sqrt{Y}, \quad X(x) = \sqrt{X},$$

$$Y = w^{2} - 2 Mw + a^{2} - n^{2} + e^{2}, \quad X = a^{2} - (n - x)^{2}$$

where σ is the energy of the Dirac particle and m_{0} is any arbitrary constant.

From equations (2.21a), (2.12g,h) and (3.3) we obtain the following equation

$$\sqrt{Y} \left[-\frac{\partial}{\partial w} + \frac{i}{Y} - \frac{Y'}{4Y} \right] \sqrt{Y} \left[\frac{\partial}{\partial w} + \frac{i}{Y} + \frac{Y'}{4Y} \right] K_{1}(w) + \frac{i}{\lambda + i} \frac{\mu_{\theta}Y}{\mu_{\theta}w} \left[\frac{\partial}{\partial w} + \frac{i}{Y} + \frac{Y'}{4Y} \right] K_{1}(w) + 2 \left(\lambda^{2} + \mu_{\theta}^{2} w^{2} \right) K_{1}(w) = 0$$
 (3.4a)

where

$$K = w^{2} \sigma - A - eQw$$

$$A = am_{o} + \sigma (a + n)^{2} \qquad ... (3.4b)$$

$$Y = w^{2} - 2Mw + a^{2} - n^{2} + e^{2}$$

and prime denotes differentiation with respect to the argument.

If w plays the role of the radial variable r, then replacing w by r and letting $K_1(r) = Y^{1/4} R(r)$ we find, after some further simplifications, that equation (3.4) becomes

-: 56 :-

-: 57 :-

.

$$\int \underline{Y} \frac{d}{dr} \left(\sqrt{\underline{Y}} \frac{dR}{dr}\right) - \frac{1}{\lambda + 1} \frac{\mu_0 \underline{Y}}{\mu_0 r} \frac{dR}{dr} + \left[\frac{K^2}{\underline{Y}} - 2\left(\lambda^2 + \mu_0^2 r^2\right)\right] \\ + \frac{K}{\lambda + 1} \frac{\mu_0 r}{\mu_0 r} + 1 \sqrt{\underline{Y}} \frac{d}{dr} \left(\frac{K}{\sqrt{\underline{Y}}}\right) R = 0 \qquad (3.5a)$$

$$K = r^2 \sigma - A - \Theta Q r \qquad (3.5b)$$

$$Y = r^2 - 2 Mr + a^2 - n^2 + e^2$$
The equation (3.5) reduces to the radial decoupled Dirac equation obtained by
$$(1) \text{ Page [93] for the Kerr - Newman black hole when n = 0,} \\ \sqrt{2} \lambda \rightarrow \lambda, \sqrt{2} \mu_0 \Rightarrow \mu_0$$

$$(1) \text{ Chandrasekhar [94] for the Kerr black hole provided} \\ e = n = 0.$$
In troducing the coordinate
$$\frac{d}{d\hat{r}} = \frac{Y}{r^2 + (a + n)^2} \frac{d}{dr} \qquad (3.6)$$

the equation (3.5) , near the horizon, reduces to

$$\frac{d^2 R}{dr^2} + (\sigma - \sigma_0)^2 R = 0 \qquad ... (3.7)$$

i.

where

.

.

$$\sigma_{0} = \frac{(a+n)^{2} \sigma + A}{r_{+}^{2} + (a+n)^{2}} + \frac{e Q r_{+}}{r_{+}^{2} + (a+n)^{2}} \qquad (3.8)$$

and r_+ called the event horizon is the positive root of Y = 0, provided $M^2 > a^2 - n^2 + e^2$. The other root of Y = 0, denoted by r_- is called the Cauchy horizon.

If n = 0, the equation (3.7) reduces to the wave equation obtained by Zhao Zheng et al. [95] for the Kerr - Newman black hole. For n = e = 0, the equation (3.7) reduces to the wave equation obtained by Liu Liao and Xu Dianyan [96] for the Kerr black hole.

.3.3 Hawking Thermal Spectrum

Now we turn to the wave equation (3.7). The solution of this equation can easily be found to be

 $R \sim \exp\left[\pm i(\sigma - \sigma_{o})\hat{r}\right] \qquad ... (3.9)$

We now write the radial wave function

 $\psi_{r} = \exp \left[-i\sigma \left(t + \hat{r}_{1} \right) \right]$... (3.10)

-: 59 :-

where

$$\hat{\mathbf{r}}_{1} = \frac{(\sigma - \sigma_{0})\hat{\mathbf{r}}}{\sigma} \qquad (3.11)$$

We resolve ψ_r into ingoing and outgoing waves as

$$\psi_r^{\text{in}} \sim \exp\left[-i\sigma(t+\hat{r}_1)\right]$$
 ... (3.12)

$$\psi_{r}^{out} \sim \exp\left[-i\sigma(t-\hat{r}_{1})\right] \qquad ... (3.13)$$

Introducing the Eddington - Finkelstein coordinates

$$\mathbf{v} = \mathbf{t} + \hat{\mathbf{r}}_1 \qquad \cdots \qquad (3.14)$$

we obtain

$$\psi_r^{\text{in}} \sim \exp(-i\sigma v)$$
 ... (3.15)

Near $r = r_{+}$, equation (3.6) can be integrated to give

$$\hat{\mathbf{r}} = \frac{1}{2\kappa_{+}} \ln(\mathbf{r} - \mathbf{r}_{+}) \qquad (3.17)$$

where

Just ouside the event horizon

$$\psi_{r}^{out} \sim e^{-i\sigma v} (r - r_{+})^{i(\sigma - \sigma_{0})/V_{+}}$$
 ... (3.19)

We now extend the outgoing wave outside the horizon to the region inside. Since on the event horizon the outgoing wave function is not analytic and can not be straightforwardly extended to the region inside, it can be continued analytically to the complex plane by going round the event horizon.

Hence inside the event horizon

$$\psi_{\vec{r}}^{\text{out}} \sim e^{-i\sigma v} (r_{+} - r_{-})^{i(\sigma - \sigma_{0})/K_{+}} e^{\pi(\sigma - \sigma_{0})/K_{+}} \dots (3.20)$$

Introducing the step function

y (x) = $\begin{cases} 1 & \text{for } x \ge 0 \\ 0 & \text{for } x < 0 \end{cases}$... (3.21)

the outgoing wave function can generally be written as

$$\phi_{r}^{out} = N_{r} [y(r - r_{+})\psi_{r}^{out}(r - r_{+}) + y(r_{+} - r)\psi_{r}^{out}(r_{+} - r) \exp \{\frac{\pi}{K_{+}}(\sigma - \sigma_{0})\}] \quad (3.22)$$

-: 61 :-

where ψ_r is the normalized Dirac wave function. Expression (3.22) describes the splitting of ϕ_r^{out} into two components :

(a) a flow of positive energy particles of strength N_r^2 outgoing from the event horizon and

(b) a flow of positive energy particles propagating in the reverse time, since inside the event horizon, r represents the time axis due to the interchange of time and space. This can be interpreted as a flow in time of negative energy particles ingoing towards the singularity region.

This shows that a wave function near the event horizon gives rise to the creation of Dirac particle-antiparticle pair [98,99].

Obviously, from the normalization condition , we obtain

$$< \phi_{r}^{out}, \phi_{r}^{out} > = N_{r}^{2} \{ \exp \left[\frac{2\pi}{\kappa_{+}} (\sigma - \sigma_{o}) \right] + 1 \} = 1$$
 ... (3.23)

or, $N_r^2 = \left\{ \exp \left[\frac{2\pi}{\kappa_+} (\sigma - \sigma_0) \right] + 1 \right\}^{-1}$

$$= \{ \exp \left[\frac{1}{K_{b}^{2}T_{+}} (\sigma - \sigma_{o}) \right] + 1 \}^{-1} \qquad \dots \qquad (3.24)$$

where

$$\Gamma_{+} = \frac{K_{+}}{2 \pi K_{b}}$$
 ... (3.25)

is the temperature near the event horizon of the NUTKN spacetime.

Equation (3.24) is the Hawking thermal spectrum formula near the event horizon of the NUTKN spacetime [100].

In the proper limits the relation (3.24) reduces to the Hawking thermal spectrum of Dirac particles near the event horizon of

Kerr - Newman black hole [101] for n = 0(1)Kerr black hole [102] if n = e = 0(ii) (iii) Reissner - Nordstrom black hole when n = a = 0Schwarzschild black hole provided n = a = e = 0(iv) NUT Kerr spacetime with **e** = 0 (v) NUT spacetime if a = e = 0 • (vi)

Thus we observe that Hawking's thermal radiation occurs not only in black hole spacetimes but also in NUT Kerr and NUT spacetimes.

3.4 Discussion

We observe particle emission near the event horizon r_+ of the NUTKN spacetime which **is not** black hole spacetime but includes all the black hole spacetimes which are asymptotically flat. The particle flux has thermal character which is evident from the formula given by (3.25). The surface gravity κ'_+ is the limit of the product of two quantities : the magnitude of proper acceleration of a stationary observer and the redshift factor. Thus particle emission can be interpreted as the escape of particles to infinity that an observer just outside the horizon sees on account of his acceleration. This fact manifests that the thermal emission is closely related to the thermal properties of the vacuum state. Finally, we conclude that Hawking's thermal radiation by black holes holds good to the spacetimes which are not black hole spacetimes but include black hole spacetimes as special cases.

-: 63 :-

-: 64 :-

CHAPTER - 4

HAWKING RADIATION OF DIRAC PARTICLES IN NUT-KERR-NEWMAN - DE SITTER SPACETIME

[103] work on thermal radiation by black holes near Hawking's black hole event horizon has been extended by Gibbons and Hawking [104] to the spacetime of cosmological event horizon . Xu Dianyan and Wang [105] and Shen You Gen [106] extended Gibbons and Hawking's Huiya work to the Kerr-Newman de Sitter spacetime . Recently Ahmed [107] carried on this work [108] to the NUT-Kerr-Newman de Sitter spacetime . Ahmed's works [109] are interesting in that Hawking's and and Hawking's thermal radiation could also be obtained in Gibbons the case of NUT de Sitter and NUT spacetimes which have peculiar properties . In this chapter we study Ahmed's work on Hawking radiation of Dirac particles near the horizons of NUT-Kerr-Newmande Sitter spacetime which includes all the black hole spacetimes (asymptotically flat or asymptotically de Sitter). We call the NUT-Kerr-Newman-de Sitter spacetime as hot NUT-Kerr-Newman (HNUTKN) spacetime since the de Sitter spacetime has been interpreted as being hot [110]. This work also provides a basis for the extension of the work on Hawking's thermal radiation by black holes near the horizons of asymptotically de Sitter background spacetimes. Using equation (2.21a) we derive the radial decoupled Dirac equation in the HNUTKN limit and then study Hawking radiation near the horizons of the HNUTKN spacetime.

4.1 The HNUTKN Spacetime

In the coordinates (u, v, w, x) the HNUTKN spacetime is given by

$$ds^{2} = \frac{x^{2} + w^{2}}{x} dx^{2} + \frac{x^{2} + w^{2}}{y} dw^{2} + \frac{x}{x^{2} + w^{2}} (du + w^{2} dv)^{2} - \frac{y}{x^{2} + w^{2}} (du - x^{2} dv)^{2} \dots (4.1a)$$

where

$$x = [a^{2} - (n - x)^{2}] [1 + \frac{\lambda}{3} (n - x^{2})] \qquad (4.1b)$$

$$Y_{-} = (w^{2} + a^{2} + n^{2}) [1 - \frac{\pi}{3}(w^{2} + 5n^{2})] - 2(Mw + n^{2}) + e^{2} \qquad (4.1c)$$

with electric potential

$$A_{\mu} d Z^{\mu} = \frac{e w}{x^2 + w^2} (du - x^2 dv) \qquad (4.2)$$

Besides the cosmological constant \wedge , the HNUTKN spacetime contains : the mass parameter M , the NUT (magnetic mass) parameter n , the angular momentum per unit mass parameter a and the electric charge parameter e. The HNUTKN spacetime includes

(i) hot Kerr - Newman spacetime for n = 0(ii) hot Kerr spacetime when n = e = 0 (iii) hot Reissner - Nordstrom spacetime for n = a = 0

(iv) hot Schwarzschild spacetime with n = a = e = 0(v) hot NUT spacetime if a = e = 0.

Thus we see that the HNUTKN spacetime includes all the black hole spacetimes (i) - (iv) which are asymptotically de Sitter. Further if we set $\Lambda = 0$, in the cases (i) - (iv), we get all the black hole spacetimes which are asymptotically flat. In the limit $\Lambda = 0$, the case (v) reduces to the NUT spacetime.

4.2 Radial Wave Equation

To obtain the HNUTKN spacetime and the radial decoupled Dirac equation we set in equations (2.1), (2.21a), (2.12g) and (2.12h) the following substitutions:

$$\epsilon_{1} = \epsilon_{2} = f = g = 1, \quad m(x) = -x^{2}, \quad p(w) = w^{2}, \quad T(w, x) = 1,$$

$$H(w) = ew, \quad G(x) = 0, \quad h(w) = w, \quad g(x) = x, \quad Z = w^{2} + x^{2},$$

$$\alpha = i\sigma, \quad \beta = -iA, \quad A = am_{o} + \sigma(a + n)^{2}, \quad X(x) = \sqrt{X},$$

$$W(w) = \sqrt{Y}, \quad X = [a^{2} - (n - x)^{2}] [1 + \frac{\Lambda}{3}(n - x)^{2}],$$

$$Y = (w^{2} + a^{2} + n^{2}) [1 - \frac{\Lambda}{3}(w^{2} + 5n^{2})] - 2(Mw + n^{2}) + e^{2}$$
(4.3)

where σ is the energy of the Dirac particle and m_{0} is any constant.

From equations (2.21a), (2.12g,h) and (4.3) we obtain after cumbersome calculations

$$\sqrt{Y} \left[-\frac{\partial}{\partial w} + \frac{i}{Y} - \frac{Y'}{4Y} \right] \sqrt{Y} \left[\frac{\partial}{\partial w} + \frac{i}{Y} + \frac{Y'}{4Y} \right] K_{1}(w)$$

$$+ \frac{i}{\lambda + i} \frac{\mu_{\Theta}}{\mu_{\Theta}} \frac{Y}{\omega} \left[\frac{\partial}{\partial w} + \frac{i}{Y} + \frac{Y'}{4Y} \right] K_{1}(w)$$

$$+ 2 \left(\lambda^{2} + \mu_{\Theta}^{2} w^{2} \right) K_{1}(w) = 0 \qquad (4.4a)$$

where

$$K = w^{2} - A - \frac{e Q w}{H}$$

$$Y = (w^{2} + a^{2} + n^{2}) \left[1 - \frac{\Lambda}{3} (w^{2} + 5n^{2})\right] - 2 (Mw + n^{2}) + e^{2}$$
(4.4b)

and prime denotes differentiation with respect to the argument.

If w plays the role of the radial variable r, then replacing w by r and letting $K_1(r) = R Y^{-1/4}$ we obtain equation (4.4) after some algebraic calculations and rearranging the terms in the following form

-: 67 :-

$$\sqrt{Y} \frac{d}{dr} \left(\sqrt{Y} \frac{d}{dr} \frac{R}{dr}\right) - \frac{i \mu_{\theta} Y}{\lambda + i \mu_{\theta} r} \frac{d}{dr} + \left[\frac{3^{2} K^{2}}{Y} - 2 \left(\lambda^{2} + \mu_{\theta}^{2} r^{2}\right) + \frac{3 K \mu_{\theta}}{\lambda + i \mu_{\theta} r} + i \frac{3}{\sqrt{Y}} \frac{d}{dr} \left(\frac{K}{\sqrt{Y}}\right)\right] R = 0 \qquad (4.5a)$$

$$K = r^{2} \sigma - A - \frac{\theta Q r}{3} \qquad (4.5b)$$

$$Y = \left(r^{2} + a^{2} + n^{2}\right) \left[1 - \frac{\Lambda}{3} \left(r^{2} + 5 n^{2}\right)\right] - 2 \left(Mr + n^{2}\right) + e^{2}$$

2

which represents the radial decoupled Dirac equation for the HNUTKN spacetime .

The equation (4.5) reduces to the radial decoupled Dirac equation obtained by

- (1) Xu Dianyan and Wang Huiya [111] and Shen You-Gen [112] for the Kerr - Newman de Sitter spacetime when n = 0
- (ii) Khanal [113] for the Kerr de Sitter spacetime if n = e = 0
- (iii) Page [114] for the Kerr Newman spacetime provided $\Lambda = n = 0$, $\sqrt{2} \lambda \rightarrow \lambda$, $\sqrt{2} \mu_{\theta} \rightarrow \mu_{\theta}$
- (iv) Chandrasekhar [115] for the Kerr spacetime when $\Lambda = e = n = 0$.

Introducing the coordinate

$$\frac{d}{d\hat{r}} = \frac{Y}{r^2 + (a+n)^2} \frac{d}{dr}$$
 (4.6)

the equation (4.5), near the horizon, reduces to

$$\frac{d^2 R}{d \hat{r}^2} + \Xi^2 (\sigma - \sigma_0)^2 R = 0 \qquad ... (4.7)$$

where

$$\sigma_{0} = \frac{(a+n)^{2}\sigma + A}{r_{+}^{2} + (a+n)^{2}} + \frac{e Q r_{+}}{\mathbf{I} [r_{+}^{2} + (a+n)^{2}]}$$

and r_+ , called the event horizon, is the smaller of the two positive values of r for which Y = 0, provided the roots are real (i.e. $\frac{1}{\Lambda} > M^2 > a^2 - n^2 + e^2$). The larger positive value of Y = 0 denoted by r_{++} represents the cosmological horizon.

If n = 0, the equation (4.7) reduces to the wave equation obtained by Xu Dianyan and Wang Huiya [116] and Shen You Gen [117] for the Kerr - Newman de Sitter spacetime. For $\wedge = n = 0$, the equation (4.7) reduces to the wave equation obtained by Zhao Zheng et al. [118] for the Kerr - Newman black hole. When $\wedge = n = e = 0$, the equation (4.7) reduces to the wave equation obtained by Liu Liao and Xu Dianyan [119] for the Kerr blak hole.

4.3 Hawking Thermal Spectrum

The solution of the equation (4.7) can easily be found to be

$$\mathbf{R} \sim \exp\left[\frac{1}{2}\mathbf{i} \equiv (\sigma - \sigma_0)\right] \hat{\mathbf{r}} \qquad \dots \qquad (4.8)$$

Now we can write the radial wave function as

$$\psi_{r} = \exp \left[-i\sigma\left(t + \hat{r}_{1}\right)\right]$$
 ... (4.9)

where

$$\hat{\mathbf{r}}_{1} = \frac{\Xi \left(\sigma - \sigma_{0}\right) \hat{\mathbf{r}}}{\sigma}$$

We resolve $\psi_{\mathbf{r}}$ into ingoing and outoing waves as

$$\psi_{r}^{111} \sim \exp\left[-i\sigma\left(t+\hat{r}_{1}\right)\right] \qquad \cdots \qquad (4.10)$$

$$\psi_{\mathbf{r}}^{\text{out}} \sim \exp\left[-\mathrm{i}\sigma\left(\mathbf{t}-\widehat{\mathbf{r}}_{1}\right)\right] \qquad \dots \qquad (4.11)$$

Introducing the Eddington - Finkelstein coordinates

we obtain

$$\psi_{\mathbf{r}}^{\text{out}} \sim \exp\left[-\mathbf{i}\,\sigma\,\mathbf{v}+2\mathbf{i}\,\Xi\,(\sigma-\sigma_{\mathbf{o}})\,\widehat{\mathbf{r}}\,\right] \qquad \dots \qquad (4.14)$$

-: 70 :-

Integrating equation (4.6), near r = r, we obtain

$$\hat{\mathbf{r}} = \frac{1}{2 \pm K_{+}} \ln (\mathbf{r} - \mathbf{r}_{+})$$
 ... (4.15)

where

$$K_{+} = - \frac{\pi}{6 \times [r_{+}^{2} + (a + n)^{2}]} (r_{+} - r_{++})(r_{+} - r_{-})(r_{+} - r_{-})$$

is the surface gravity [120] of the event horizon of the HNUTKN spacetime. Here r_{is} is the inner black hole horizon and r_{is} is enother cosmological horizon.

Just outside the event horizon we have

$$\psi_{-}^{\text{out}} \sim e^{-i\sigma v} (r - r_{+})^{(i/K_{+})(\sigma - \sigma_{0})} \dots (4.16)$$

We now extend the outgoing wave outside the horizon to the region inside . Since on the event horizon, the outgoing wave function is not analytic and can not be straightforwardly extended to the region inside, it can be continued analytically to the complex plane by going round the event horizon .

Hence inside the event horizon

$$\psi_{r}^{\text{out}} \sim e^{-i\sigma v} (r_{+} - r)^{(i/K_{+})(\sigma - \sigma_{0})} e^{(\pi/K_{+})(\sigma - \sigma_{0})} \cdots (4.17)$$

Introducing the step function

$$y(x) = \begin{cases} 1 , x \ge 0 \\ 0 , x < 0 \end{cases}$$
 (4.18)

the outgoing wave function can generally be written as

$$\phi_{\mathbf{r}}^{\text{out}} = N_{\mathbf{r}} \left[y \left(\mathbf{r} - \mathbf{r}_{+} \right) \psi_{\mathbf{r}}^{\text{out}} \left(\mathbf{r} - \mathbf{r}_{+} \right) + y \left(\mathbf{r}_{+} - \mathbf{r} \right) \psi_{\mathbf{r}}^{\text{out}} \left(\mathbf{r}_{+} - \mathbf{r} \right) \exp \left\{ \frac{\pi}{\kappa_{+}} \left(\sigma - \sigma_{0} \right) \right\} \right] \dots (4.19)$$

where ψ is the normalized Dirac wave function .

Expression (4.19) describes the splitting of ϕ_r^{out} into two components :

(a) a flow of positive energy particles of strength N_r^2 outgoing from the event horizon and

(b) a flow of positive energy particles propagating in the HNUTKN background gravitational field in the reverse time, since inside the event horizon, r represents the time axis due to the interchange of time and space. This can be interpreted as a flow in time of negative energy antiparticles ingoing towards the singularity region. This shows that a wave function near the event horizon gives rise to the creation of Dirac particle antiparticle pair [121]. From the normalization condition , we have

$$\langle \phi_r^{\text{out}}, \phi_r^{\text{out}} \rangle = N_r^2 \{ \exp \left[\frac{2 \pi}{\kappa_+} (\sigma - \sigma_0) \right] + 1 \} = 1$$
 (4.20)

or $N_r^2 = \{ \exp \left[\frac{2\pi}{\kappa_+} (\sigma - \sigma_0) \right] + 1 \}^{-1}$

$$= \left\{ \exp \left[\frac{1}{\kappa_{b}^{T} T_{+}} (\sigma - \sigma_{o}) \right] + 1 \right\}^{-1} \qquad (4.21)$$

where

$$T_{+} = \frac{K_{+}}{2 \pi K_{h}}$$
 (4.22)

 T_+ being the temperature of the region inside the event horizon, K_b is the Boltzmann's constant . Equation (4.21) is the formula for the Hawking thermal spectrum of Dirac particles in the HNUTKN spacetime [122].

In the limits

(i) n = 0, the relation (4.21) will give the Hawking thermal spectrum of Dirac particles in HKN spacetime [123] (ii) n = e = 0, we get from (4.21) the Hawking thermal spectrum of Dirac particles in the hot Kerr spacetime (iii) n = a = 0, the relation (4.21) gives the Hawking thermal spectrum in hot Reissner - Nordstrom spacetime

(iv) n = a = e = 0, the relation (4.21) is the Hawking thermal spectrum in the hot Schwarzschild spacetime.

Further in the limits $\Lambda = 0$, in the cases (i) - (vi) we will get the Hawking thermal spectrum formula for

(i) Kerr - Newman spacetime [124]

(ii) Kerr spacetime [125]

(iii) Reissner Nordstrom spacetime

(iv) Schwarzschild spacetime .

In the limit a = e = 0, the relation (4.21) will give the formula for the Hawking thermal spectrum in the hot NUT spacetime. Further in the limit $\wedge = 0$, we get the formula for the Hawking thermal spectrum in the NUT spacetime.

Following in the similar way we could have

$$\Gamma_{++} = \frac{\kappa_{++}}{2\pi \kappa_{\rm b}} \qquad \dots \qquad (4.23)$$

where

$$\mathcal{K}_{++} = - \frac{\Lambda}{6 \equiv [r_{++}^2 + (a+n)^2]} (r_{++} - r_{+}) (r_{++} - r_{-}) (r_{++} - r_{-}) (4.24)$$

is the surface gravity of the cosmological horizon . In the proper limits (4.24) goes for the results obtained in [126].

4.4 Discussion

We observe particle emission near the horizons r_+ and r_{++} of the HNUTKN spacetime which is not a black hole spacetime but includes all the black hole spacetimes which are asymptotically flat as well as asymptotically de Sitter as special cases. In the next chapter we shall show that Gibbons and Hawking's thermal radiation by black holes occurs in the case of Kasner - type spacetime which describes an anisotropic model of the universe. -: 76 :-

CHAPTER -5

HAWKING RADIATION OF DIRAC PARTICLES IN KASNER- TYPE SPACETIME

Hawking's [127] investigations of quantum effects interpreted as the emission of a thermal spectrum of particles near a black hole event horizon has been extended by Gibbons and Hawking [128] to the spacetime of cosmological event horizons including the de Sitter spacetime, which has attracted renewed interest as a model of the early universe . In Chapter 3, we have observed that Hawking's [129] result on thermal radiation by black holes holds good in the case of NUT - Kerr - Newman spacetime which includes all the black hole spacetimes (asymptotically flat) as well as NUT spacetime as special cases. Whereas in Chapter 4, we have observed that Gibbons and Hawking's [130] result holds good in the case of NUT - Kerr - Newman - de Sitter spacetime which includes black hole spacetimes (asymptotically flat as well as asymptotically de Sitter) and NUT de Sitter spacetime as special cases. In the present chapter we have attempted to show that Hawking's [131] and Gibbons and Hawking's [132] results on the thermal radiation by black holes that have been found to occur in the case of NUT - Kerr - Newman and NUT - Kerr - Newman - de Sitter spacetimes also occur in the case of Kasner - type spacetime , an anisotropic universe .

-: 77 :-

5.1 The Kasner - Type Spacetime

We consider the spacetime

$$ds^{2} = r^{2} (d\theta^{2} + \theta^{2} d\phi^{2}) + \frac{dr^{2}}{\Delta} - \Delta dt^{2} \qquad (5.1a)$$

where

$$\Delta = -\frac{1}{3} \wedge r^2 - \frac{2M}{r} + \frac{e^2 + g^2}{r^2} \qquad (5.1b)$$

Besides the cosmological constant \wedge , the spacetime given by (5.1) contains three real parameters : the mass M, the electric charge e and the magnetic charge g.

The spacetime (5.1) can be transformed to the Kasner form when we put $\Lambda = e = g = 0$. Equation (5.1) can be written in the form

$$ds^{2} = r^{2} (d\theta^{2} + \theta^{2} d\phi^{2}) + \frac{r^{2}}{r} dr^{2} - \frac{Y}{r^{2}} dt^{2} \qquad (5.2a)$$

where

$$Y = -\frac{\Lambda}{3}r^4 - 2Mr + e^2 + g^2 \qquad ... (5.2b)$$

The surfaces Y = 0 are the horizons of the metric (5.2).

-: 78 :-

5.2 Radial Wave Equation

í.

From equations (2.21a), (2.12g,h) in the limit, we obtain the radial (w \sim r) decoupled Dirac equation in the case of Kasner - type spacetime as follows :

$$Y \frac{d^{2}R}{dr^{2}} + \left[\sqrt{Y} \frac{d}{dr} \left(\sqrt{Y}\right) - \frac{i \mu_{\theta} Y}{\lambda + i \mu_{\theta} r}\right] \frac{d R}{dr} + \left[\frac{K^{2}}{Y} - 2\left(\lambda^{2} + \mu_{\theta}^{2} r^{2}\right)\right]$$

$$+ \frac{\mu_{\Theta}K}{\lambda + i\mu_{\Theta}r} + i\sqrt{Y} \frac{d}{dr} \left(\frac{K}{\sqrt{Y}}\right) = 0 \qquad ... (5.3)$$

$$\mathbf{K} = \mathbf{r}^2 \mathbf{\sigma} - \mathbf{e} \mathbf{Q} \mathbf{r}$$

$$Y = -\frac{\Lambda}{3}r^4 - 2Mr + e^2 + g^2$$

where σ is the energy of the Dirac particle, λ is the separation constant, μ_e and Q are the mass and the electric charge of the Dirac particle respectively.

With the coordinate transformation

$$\frac{d}{d\hat{r}} = \frac{Y}{r^2} \frac{d}{dr}$$
 (5.4)

equation (5.3) reduces near the horizon to the form

-: 79 :-

$$\frac{d^{2}R}{d\hat{r}^{2}} + (\sigma - \sigma_{0})^{2} R = 0 \qquad ... (5.5)$$

where

$$\sigma_{0} = \frac{\Theta Q}{r_{+}} \qquad \dots \qquad (5.6)$$

and r_+ , called the event horizon, is the smaller of the two positive values of r at which Y = 0, provided the roots are real. The larger positive value of Y = 0, denoted by r_{++} , represents the cosmological horizon.

5.3 Hawking Thermal Spectrum

The solution of equation (5.5) can easily be found to be

$$\mathbf{R} \sim \exp \left[\frac{1}{2} \mathbf{i} \left(\sigma - \sigma_{o} \right) \right] \hat{\mathbf{r}} \qquad \dots \qquad (5.7)$$

Now we can write the radial wave function as

$$\Psi_{r} = \exp \left[-i\sigma \left(t + \hat{r}_{1} \right) \right]$$
 ... (5.8)

where

$$\hat{\mathbf{r}}_{1} = \frac{\sigma - \sigma_{0}}{\sigma} \hat{\mathbf{r}} \qquad \dots \qquad (5.9)$$

We resolve
$$\psi_r$$
 into ingoing and outgoing waves as

$$\psi_{\mathbf{r}}^{\text{in}} \sim \exp\left[-\mathbf{i}\sigma\left(\mathbf{t}+\hat{\mathbf{r}}_{1}\right)\right] \qquad \cdots \qquad (5.10)$$

$$\psi_{\mathbf{r}}^{\text{out}} \sim \exp\left[-\mathbf{i}\sigma\left(\mathbf{t}-\hat{\mathbf{r}}_{1}\right)\right] \qquad \cdots \qquad (5.11)$$

Introducing the Eddington - Finkelstein coordinates

$$\mathbf{v} = \mathbf{t} + \hat{\mathbf{r}}_1 \qquad \dots \qquad (5.12)$$

we obtain

$$\psi_r^{\text{in}} \sim \exp(-\operatorname{i}\sigma v) \qquad \dots \qquad (5.13)$$

$$\psi_{\mathbf{r}}^{\text{out}} \sim \exp\left[-\operatorname{i}\sigma \mathbf{v} + 2\mathbf{i}(\sigma - \sigma_{0})\mathbf{\hat{r}}\right] \qquad \dots \qquad (5.14)$$

Near $r = r_{+}$, equation (5.4) can be integrated to give

$$\hat{\mathbf{r}} = \frac{1}{2\kappa_{\perp}} \ln(\mathbf{r} - \mathbf{r}_{\perp})$$
 ... (5.15)

where

$$\mathcal{K}_{+} = - \frac{\Lambda}{6r_{+}^{2}} (r_{+} - r_{++})(r_{+} - r_{-})(r_{+} - r_{--}) \qquad (5.16)$$

is the surface gravity of the event horizon of the Kasner-type spacetime. Here r_{is} is the inner black hole horizon and r_{is} is another cosmological horizon.

Just outside the event horizon we have

$$\psi_r^{\text{out}} \sim e^{-i\sigma v} (r - r_+)^{(i/K_+)(\sigma - \sigma_0)} \qquad \dots \qquad (5.17)$$

We now extend the outgoing wave outside the horizon to the region inside. Since on the event horizon the outgoing wave function is not analytic and cannot be straightforwardly extended to the region inside , it can be continued analytically to the complex plane by going around the event horizon.

Hence inside the event horizon

$$\psi_{\mathbf{r}}^{\text{out}} \sim e^{-\mathbf{i} \sigma \mathbf{v}} (\mathbf{r}_{+} - \mathbf{r})^{(\mathbf{i}/K_{+})(\sigma - \sigma_{0})} e^{(\pi/K_{+})(\sigma - \sigma_{0})}$$
(5.18)

Introducing the step function

$$y(x) = \begin{cases} 1 & \text{for } x \ge 0 \\ 0 & \text{for } x < 0 \end{cases}$$
 (5.19)

we can generally write the outgoing wave function as

$$\phi_{\mathbf{r}}^{\text{out}} = \mathbb{N}_{\mathbf{r}} \left\{ \mathbf{y}(\mathbf{r} - \mathbf{r}_{+}) \psi_{\mathbf{r}}^{\text{out}} (\mathbf{r} - \mathbf{r}_{+}) \right\}$$

+y(r₊ - r)
$$\psi_r^{out}$$
 (r₊ - r) exp [$\frac{\pi}{\kappa_+}$ ($\sigma - \sigma_0$)]} ... (5.20)

where ψ_r^{out} is the normalized Dirac wave function.

-: 81 :-

-: 82 :-

Expression (5.20) describes the splitting of ϕ_r^{out} into two components :

(a) A flow of positive - energy particles of strength N_r^2 outgoing from the event horizon .

(b) A flow of positive - energy particles in the Kasner-type background in the reverse time, since inside the event horizon, r represents the time axis due to the interchange of time and space. This can be interpreted as a flow in time of negative - energy particles ingoing toward the singularity region. This shows that a wave function near the event horizon gives rise to the creation of a Dirac particle - antiparticle pair [133].

Obviously, from the normalization condition we have

$$<\phi_{r}^{\text{out}}, \phi_{r}^{\text{out}} > = N_{r}^{2} \{ \exp \left[\frac{2 \pi}{K_{+}} (\sigma - \sigma_{0}) \right] + 1 \} = 1 \quad ... \quad (5.21)$$

or, $N_{r}^{2} = \{ \exp \left[\frac{2 \pi}{K_{+}} (\sigma - \sigma_{0}) \right] + 1 \}^{-1}$
$$= \{ \exp \left[\frac{1}{K_{b}} T_{+} (\sigma - \sigma_{0}) \right] + 1 \}^{-1} \quad ... \quad (5.22)$$

where

$$T_{+} = \frac{K_{+}}{2 \pi K_{b}} \qquad (5.23)$$

T₊ is the temperature of the region inside the event horizon and \mathcal{K}_b is Boltzmann's constant. Equation (5.22) is the formula for the Hawking thermal spectrum of Dirac particles in the Kasner - type spacetime [134].

Following in a similar way , we have

$$T_{++} = \frac{K_{++}}{2 \pi K_{b}}$$
 (5.24)

where

$$\mathcal{K}_{++} = - \frac{\Lambda}{6r_{++}^2} (r_{++} - r_{+})(r_{++} - r_{-})(r_{++} - r_{--}) \dots (5.25)$$

is the surface gravity of the cosmological event horizon .

5.4 Remarks

From this work it has appeared that an anisotropic model like Kasner-type spacetime gives rise to particle creation .The Kasnertype spacetime is due to the contraction of the Schwarzschild spacetime generalized with the cosmological constant and electric and magnetic monopole charge. This result of particle creation in the Kasner - type spacetime goes beyond the idea that in the contraction phase it is necessary that the matter should disappear. To avoid this situation, it could be said that particles do not disappear in the process of contraction but become immaterial. It will be more interesting to say that " immaterial souls " of particles are created during contraction.

CHAPTER - 6

HAWKING RADIATION OF DIRAC PARTICLES IN PLEBANSKI SPACETIME

In the preceding three chapters we have seen that Hawking's and Gibbons and Hawking's thermal spectrum of Dirac particles near the event horizon and the cosmological horizon of black hole spacetimes holds good near the event horizon and the cosmological horizon of NUTKN , HNUTKN and Kasner- type spacetimes . In the present chapter we shall generalize these results and show that Hawking's and Gibbons and Hawking's thermal radiation by black holes holds good also in a more general background spacetime viz. Plebanski spacetime . Using equation (2.21a) we derive the radial decoupled Dirac equation in the Plebanski limit and then study Hawking radiation near the horizons of the metric .

6.1 The Plebanski Spacetime

In the coordinates (u, v, w, x) the Plebanski spacetime is given by

$$ds^{2} = \frac{x^{2} + w^{2}}{x} dx^{2} + \frac{x^{2} + w^{2}}{y} dw^{2} + \frac{x}{x^{2} + w^{2}} (du + w^{2} dv)^{2}$$

$$- \frac{y}{x^{2} + w^{2}} (du - x^{2} dv)^{2} \qquad (6.1a)$$

where

$$X = b - g^{2} + 2n x - \varepsilon x^{2} - \frac{h}{3} x^{4} \qquad (6.1b)$$

$$Y = b + e^{2} - 2Mw + \varepsilon w^{2} - \frac{h}{3} w^{4} \qquad (6.1c)$$

with electric potential

$$A_{\mu} dx^{\mu} = \frac{ew}{x^2 + w^2} (du - x^2 dv) \qquad (6.2)$$

Besides the cosmological constant \wedge , the metric given by (6.1) includes : the mass parameter M, the NUT (magnetic mass) parameter n, the electric charge parameter e, the magnetic charge parameter g and the kinetical parameters **b** and ε . The metric is of Petrov type - D and the surfaces Y = 0are the horizons of the metric [135].

With suitable adjustment of the parameters and appropriate coordinate transformation the metric (6.1) gives many of the physically interesting solutions of Einstein or Einstein -Maxwell equation which has already been discussed in Chapter 1.

6.2 Radial Wave Equation

To obtain the Plebanski spacetime and the radial decoupled Dirac equation in this background spacetime, we set in equations (2.1), $(2.21a)_{,}(2.12g)$ and (2.12h) the following substitutions:

$$\epsilon_{1} = \epsilon_{2} = f = g = 1, \quad m(x) = -x^{2}, \quad p(w) = w^{2}, \quad T(w, x) = 1,$$

$$H(w) = ew, \quad G(x) = 0, \quad h(w) = w, \quad g(x) = x, \quad Z = w^{2} + x^{2},$$

$$W(w) = \sqrt{Y}, \quad X(x) = \sqrt{X}, \quad \alpha = i \sigma, \quad \beta = -i \Lambda, \quad (6.3)$$

$$Y = b + e^{2} - 2Mw + \epsilon w^{2} - \frac{\Lambda}{3}w^{4},$$

$$X = b - g^{2} + 2nx - \epsilon x^{2} - \frac{\Lambda}{3}x^{4}$$

where σ is the energy of the Dirac particle and A is the arbitrary constant.

From equations (2.21a), (2.12g, h) and (6.3) we obtain the following equation

$$\sqrt{Y} \left[-\frac{\partial}{\partial w} + \frac{i}{Y} - \frac{Y'}{4Y} \right] \sqrt{Y} \left[\frac{\partial}{\partial w} + \frac{i}{Y} + \frac{Y'}{4Y} \right] K_{1}(w)$$

$$+ \frac{i}{\lambda + i} \frac{\mu_{\theta}}{\mu_{\theta}} \left[\frac{\partial}{\partial w} + \frac{i}{Y} + \frac{Y'}{4Y} \right] K_{1}(w)$$

$$+ 2 \left(\frac{\lambda^{2}}{\lambda} + \frac{\mu^{2}}{2} \frac{w^{2}}{2} \right) K_{1}(w) = 0 \qquad (6.4)$$

where

$$K = w^{2} \sigma - A - eQw$$

$$Y = b + e^{2} - 2Mw + ew^{2} - \frac{\Lambda}{3}w^{4}$$

If w plays the role of the radial variable r, then replacing w by r and letting $K_1(r) = RY^{-1/4}$ we obtain (6.4) as

$$\sqrt{Y} \frac{d}{dr} \left(\sqrt{Y} \frac{dR}{dr}\right) - \frac{i\mu_{\theta}Y}{\lambda + i\mu_{\theta}r} \frac{dR}{dr} + \left[\frac{K^2}{Y} - 2\left(\lambda^2 + \mu_{\theta}^2 r^2\right)\right]$$

$$+ \frac{K \mu_{\Theta}}{\lambda + i \mu_{\Theta} r} + i \sqrt{Y} \frac{d}{dr} \left(\frac{K}{\sqrt{Y}}\right) R = 0 \qquad (6.5)$$

$$K = r^{2} \sigma - A - eQr$$

$$Y = b + e^{2} - 2Mr + \epsilon r^{2} - \frac{\Lambda}{--} r^{4}$$

Under the coordinate transformation (1.42) along with the adjustment of the parameters (1.39) and (1.40) the equation (6.5) transforms to the radial decoupled Dirac equation for the NUT-Kerr-Newman-Kasuya-de Sitter spacetime which reduces to the radial decoupled Dirac equation obtained by

- (1) Ahmed [136] for the HNUTKN spacetime when g = 0
- (ii) Xu Dianyan and Wang Huiya [137] for the Kerr-Newman de Sitter spacetime if n = g = 0
- (iii) Khanal [138] for the Kerr- de Sitter spacetime provided n = e = g = 0

-: 88 :-

- (iv) Page [139] for the Kerr Newman spacetime when $\sqrt{2} \lambda \rightarrow \lambda$, $\sqrt{2} \mu_{\Theta} \rightarrow \mu_{\Theta}$, $\Lambda = n = g = 0$
- (v) Chandrasekhar [140] for the Kerr spacetime if $\Lambda = e = n = g = 0$.

Introducing the coordinate $\ \hat{\mathbf{r}}$ defined by

$$\frac{d}{d\hat{r}} = \frac{Y}{r^2 + x^2} \frac{d}{dr} \qquad (6.6)$$

the equation (6.5) near the horizon Y = 0, reduces to the following form

$$\frac{d^2 R}{d\hat{r}^2} + (\sigma - \sigma_0)^2 R = 0 \qquad ... (6.7)$$

where

$$\sigma_{o} = \left[\frac{x^{2}\sigma + A}{r^{2} + x^{2}} + \frac{eQr}{r^{2} + x^{2}}\right]_{r=r_{+}} \qquad (6.8)$$

and r_+ , called the event horizon, is one of the positive values of r for which Y = 0, provided the roots of Y = 0 are real.

Under the coordinate transformation (1.42) accompanied by the adjustment of the parameters (1.39) and (1.40), the equation (6.7) will be transformed to the wave equation for the HNUTKNK spacetime.

-: 89 :-

6.3 Hawking Thermal Spectrum

The solution of equation (6.7) is evidently

$$\mathbf{R} \approx \exp\left[\frac{1}{2}\mathbf{i}\left(\sigma - \sigma_{0}\right)\right]\hat{\mathbf{r}} \qquad \dots \qquad (6.9)$$

Now we write the radial wave function

$$\psi_{\mathbf{r}} = \exp \left[-i\sigma \left(t + \hat{r}_{1} \right) \right]$$
 ... (6.10)

where

-

We resolve ψ_r into ingoing and outgoing waves as

$$\psi_{r}^{in} \sim \exp\left[-i\sigma\left(t+\hat{r}_{1}\right)\right]$$
 ... (6.12)

$$\psi_{\mathbf{r}}^{\text{out}} \approx \exp\left[-\mathrm{i}\sigma\left(\mathbf{t} - \mathbf{r}_{1}\right)\right] \qquad \dots \qquad (6.13)$$

Introducing the Eddington - Finkelstein coordinates

$$v = t + \hat{r}_1$$
 (6.14)

we obtain

$$\psi_{\mathbf{r}}^{\text{out}} \sim \exp\left[-\mathbf{i}\sigma \mathbf{v} + 2\mathbf{i}(\sigma - \sigma_{o})\hat{\mathbf{r}}\right]$$
 ... (6.16)

-: 90 :-

Now on integration of equation (6.6), we find near r = r + that the equation takes the following form

$$\hat{\mathbf{r}} = \frac{1}{2\kappa_{+}} \ln(\mathbf{r} - \mathbf{r}_{+})$$
 ... (6.17)

where

$$\mathcal{K}_{+} = \frac{1}{2} \begin{bmatrix} 1 & 0 & Y \\ r^{2} + x^{2} & -r^{2} \end{bmatrix}_{\mathbf{r}} = \mathbf{r}_{+}$$
 (6.18)

is the surface gravity [141] near the event horizon of the Plebanski spacetime .

Just outside the event horizon

$$\psi_{\mathbf{r}}^{\text{out}} \sim e^{-i\sigma \mathbf{v}} (\mathbf{r} - \mathbf{r}_{\mathbf{i}})^{(i/\mathcal{K}_{\mathbf{i}})(\sigma - \sigma_{o})} \qquad \dots \qquad (6.19)$$

By analytically continuing the outgoing wave outside the horizon to the region inside we obtain inside the event horizon

$$\psi_{r}^{\text{out}} \sim e^{-i\sigma V} (r_{+} - r_{-})^{(i/K_{+})(\sigma - \sigma_{0})} e^{(\pi/K_{+})(\sigma - \sigma_{0})} \dots (6.20)$$

Introducing the step function

$$y(x) = \begin{cases} 1 & \text{for } x \ge 0 \\ 0 & \text{for } x < 0 \end{cases}$$
 (6.21)

the outgoing wave function can generally be written as

$$\phi_{r}^{\text{out}} = N_{r} [y(r - r_{+}) \psi_{r}^{\text{out}} (r - r_{+}) + y(r_{+} - r) \psi_{r}^{\text{out}} (r_{+} - r) \exp \{\frac{\pi}{K_{+}} (\sigma - \sigma_{0})\}] \dots (6.22)$$

where ψ_r^{out} is the normalized Dirac wave function. The above expression shows that a wave function near the event horizon gives rise to the creation of Dirac particle antiparticle pair [142].

From the normalization condition , we obtain

$$\langle \phi_{r}^{\text{out}}, \phi_{r}^{\text{out}} \rangle = N_{r}^{2} \{ \exp \left[\frac{2\pi}{K_{+}} (\sigma - \sigma) \right] + 1 \} = 1...$$
 (6.23)

or, $N_r^2 = \{ \exp \left[\frac{2\pi}{K_+} (\sigma - \sigma_0) \right] + 1 \}^{-1}$

$$= \{ \exp \left[\frac{1}{K_{b} T_{+}} (\sigma - \sigma_{o}) \right] + 1 \}^{-1} \qquad ... \quad (6.24)$$

where

$$\mathbf{T}_{+} = \frac{K_{+}}{2 \pi K_{b}} \qquad (6.25)$$

being the temperature of the region inside the event horizon and K_b is the Boltzmann's constant. Equation (6.24) is

-: 92 :-

the formula for the Hawking thermal spectrum of Dirac particles near the event horizon of the Plebanski spacetime .

Under the coordinate transformation (1.42) accompanied by the adjustment of the parameters (1.39) and (1.40), the equations (6.5) - (6.25), will be transformed to the corresponding equations of HNUTKNK spacetime [143]. The Hawking thermal spectrum formula thus obtained will be reduced to the Hawking thermal spectrum formula of Dirac particles near the event horizons of

(i) HKNK spacetime when
$$n = 0$$

(ii) HNUTKN spacetime [144] for g = 0

(iii) hot Kerr - Newman spacetime [145] for n = g = 0

(iv) hot Kerr - spacetime if n = g = e = 0

(v) hot Reissner - Nordstrom spacetime provided n = g = a = 0

(vi) hot Schwarzschild spacetime for n = g = a = e = 0

(vii) hot NUT spacetime when $\mathbf{a} = \mathbf{e} = \mathbf{g} = \mathbf{0}$.

Further in the limits $\Lambda = 0$ in the cases (i) - (vii), we will get the Hawking thermal spectrum for

(i) Kerr - Newman-Kasuya [146] spacetime

-1 73 1-

-: 94 :-

- NUT Kerr Newman [147] spacetime (ii)
- Kerr Newman [148] spacetime (111)
- [149] spacetime Kerr (iv)
- Reissner Nordstrom spacetime (v)
- Schwarzschild spacetime (vi)
- (vii) NUT spacetime .

Following the similar way, we could have

$$T_{++} = \frac{K_{++}}{2 \pi K_b}$$
 (6.26)

where

$$\mathbf{K}_{++} = \frac{1}{2} \begin{bmatrix} \frac{1}{r^2 + x^2} & \frac{\partial Y}{\partial r} \end{bmatrix}_{r=r} + +$$
(6.27)

r_+ , called the cosmological horizon , is the larger and positive root of Y = 0 and K_{++} represents the surface gravity [150] of the cosmological horizon of the Plebanski spacetime .

Under the coordinate transformation (1.42) accompanied by the adjustment of the parameters (1.39) and (1.40), the . equation (6.27) will be transformed to the corresponding

equation for the HNUTKNK spacetime [151] which with proper limits goes for the results obtained in [152].

Thus we observe that our result is a more general one and encompasses all the results obtained in [153].

6.4 Discussion

We observe that Hawking-Gibbon's thermal spectrum formula of Dirac particles near the event horizon and the cosmological horizon of the black hole spacetimes, can also be obtained near the event horizon and the cosmological horizon of the more general spacetime such as Plebanski spacetime which includes many interesting spacetimes besides the black hole spacetimes as special cases. So we see that the spacetimes which are not black hole spacetimes but includes black hole spacetimes having horizons can produce thermal particles near the horizons. The Hawking thermal spectrum formula obtained in the Plebanski background spacetime will be reduced to all the particular cases of the Plebanski spacetime . From the observation of the particular cases of the Plebanski spacetime which has already been discussed in Chapter 1, we see that the Hawking radiation is valid for all types of spacetimes having horizons . The result goes most of the way in confirming the theory obtained recently by Kay and Wald [154] that Hawking radiation is valid for the arbitrary spacetime having killing horizon.

-: 95 :-

-: 96 :-

REFERENCES

i.

1.	E.Schrodinger, Physica <u>6</u> , 899 (1939)
2.	L.Rosenfeld, Ann. Physik <u>5</u> , 133 (1930); Z Physik <u>65</u> , 589 (1930)
3.	L. Parker, Phys. Rev. Lett. 21, 562 (1968)
4.	R.U. Sexl and H.K. Urbantke , Acta Phys. Austriaca 26 , 339 (1967)
5.	S.W. Hawking, Nature , 248, 30 (1974)
6.	S.W. Hawking, Commun . Math. Phys. <u>43</u> , 199 (1975)
7.	J. Barden et al., Commun. Math. Phys. <u>31</u> , 161 (1973)
8.	G.W. Gibbons and S.W. Hawking, Phys. Rev. D 15, 2738(1977)
9.	Reference 6
10.	J.D. Bekenstein, Phys. Rev. D 12, 3077 (1975)
11.	S.W. Hawking, in " General Relativity : An Einstein Centenary Survey " edited by S.W. Hawking and W. Israel (Cambridge University Press, Cambridge, 1979)
12.	D.J. Gross, M.J. Perry and L.G. Yaffe Phys. Rev. D 25, 330 (1982)
13.	Liu Liao and Xu Dianyan , Acta Phys.Sin. 29 , 1617 (1980)
14.	Zhao Zheng et al., Acta Astrophys.Sin. 1, 141 (1981)
15.	Xu Dianyan and Wang Huiya, Acta Scie.Nat. 5, 63 (1985)
16.	Shen You-Gen, Acta Phys. Sin. <u>34</u> , 1202 (1985)
17.	M. Ahmed, Class. Quantum Grav. 4, 431 (1987)
18.	M. Ahmed , Phys. Lett. B 258 , 318 (1991)
19.	M. Ahmed and A.K. Mondal, Int. J. Theo. Phys. 31, 2047 (1992)
20,	References 17, 18
21.	Reference 19

-: 97 :-

.

	22. J.F. Plebanski, Ann. Phys. <u>90</u> , 196 (1975)
ŝ	23. N. Kamran and R.G. McLenaghan , J. Math. Phys. 25 ,1019 (1984)
	24. S. Chandrasekhar, Proc. Roy. Soc. London A 349, 571 (1976)
	25. S. Chandrasekhar, The Mathematical Theory of Black Holes (Oxford Clarendon, 1987) P. 546
	26. D.N. Page , Phys. Rev. D 14 , 1509 (1976)
	27. References 13 - 19
	28. Reference 22
	29. Reference 22, Eqn. (12.17)
•	30. M. Kasuya, Phys. Rev. D 25, 995 (1982)
	31. M. Demianski and E.T. Newman , Bull. Acad. Pol. Sci. Ser. Math. Astron. Phys. <u>14</u> , 653 (1966)
	32. E.T. Newman et al., J. Math. Phys. <u>6</u> , 918 (1965)
	33. R.P. Kerr , Phys. Rev. Lett. <u>11</u> , 237 (1963)
	34. G. Nordstrom , Proc. Kon. Ned. Akad. Wet. 20 , 1238 (1918)
	35. H. Reissner, Ann. Phys. 50, 106 (1916)
	36. K. Schwarzschild , Berliner Sitzungsler 189 (1916)
	37. D. Brill, Phys. Rev. B 133, 845 (1964)
	38. E.T. Newman et al., J.Math. Phys. <u>4</u> , 915 (1963)
	39. M. Ahmed and A.K. Mondal, To be submitted for publication
	40. Reference 39
	41. M. Gasperini, Class. Quantum Grav. 5, 521 (1988)
	42. Reference 8; B. Carter, Commun. Math. Phys. 17, 233 (1970)
	43. Reference 8
	44. Reference 8

.

•

.

.

.

-: 98 :-

45. M. Demianski, Phys. Lett. A <u>42</u> , 157 (1972)
46. C.W. Misner, J. Math. Phys. <u>4</u> , 924 (1963)
47. A.H. Guth, Phys. Rev. D 23, 347 (1981)
48. M. Ahmed and A.K. Mondal, To be submitted for publication
49. Reference 22
50. B. Carter, Phys. Rev. <u>174</u> , 1559 (1968)
51. B. Carter, Commun. Math. Phys. <u>10</u> , 280 (1968)
52. R. Debever et al., J. Math. Phys. 25, 1955 (1984)
53. S.A. Teukolsky, J. Astrophys. <u>185</u> , 635 (1973)
54. J.F. Plebanski and M. Demianski, Ann. Phys. <u>98</u> , 98 (1976)
55. A.L. Dudley and J.D. Finley III, J. Math. Phys. 20, 311 (1979)
56. Reference 24
57. Reference 26
58. B. Carter and R.G. McLenaghan, Phys. Rev. D <u>19</u> , 1093 (1979)
59. B. Carter and R.G. McLenaghan , in"Proceedings of the Second Marcel Grossman Meeting on General Relativity" edited by R. Ruffini (North-Holland Publishing Company , 1982).
60. Reference 23
61. Reference 52
62. Reference 26
63. Reference 24
64. R. Debever and R.G. McLenaghan, J.Math. Phys. 22, 1711 (1981)
65. R. Debever and R.G. McLenaghan, Bull. Cl. Sci. Acad. R. Belg. <u>LXVI</u> , 204 (1980)
66. A. G. Diaz and Salazar I, J. Math. Phys. <u>24</u> , 2498 (1983)
67. R. Debever et al., J. Math. Phys. 25, 1955 (1984)

•

÷.,

68. R. Debever et al., Phys. Lett. A 93, 399 (1983) 69. A.G. Diaz, J. Math. Phys. 25, 1951 (1984) 70. Reference 54 71. Reference 22 72. W. Kinnersley, J.Math. Phys. <u>10</u>, 1195 (1969) 73. B. Carter, Phys. Lett. A 26, 399 (1968) 74. B. Carter, Commun. Math. Phys. 10, 280 (1968) 75. B. Carter , in " Black Holes ", edited by B. De Witt and C.De Witt (Gordan and Breach, New York, 1973) P. 89 76. Reference 52, Eqn. (2.5) 77. R. Debever et al., Bull. Cl. Sci. Acad. R. Belg. LXVIII , 592 (1982) 78. Reference 24 79. Reference 26 80. R. Guven, Proc. Roy. Soc. London A 356, 465 (1977) 81. Reference 24 24. 82. Reference 26 83. Reference 24 84. Reference 85. Reference 53 5,6 86. References 24 87. Reference 88. Reference 13 26 89. Reference 90. Reference 14 17 91. Reference 23 92. Reference

-: 99 :-

- 93. Reference 26
- 94. Reference 24
- 95. Reference 14
- 96. Reference 13
- 97. R.M. Wald, in "Black holes, Thermodynamics and Time reversibility " (Preprint, 1980, Enrico, Fermi Institute, Chicago, Illinois, U.S.A.)
- 98. N. Deruelle and R. Ruffini, Phys. Lett. B 52, 437 (1975)
- 99. N. Deruelle and R. Ruffini, Phys. Lett. B 57, 248 (1975)
- 100. Reference 17
- 101 Reference 14
- 102 Reference 13
- 103. References 5,6
- 104. Reference 8
- 105. Reference 15
- 106. Reference 16
- 107. Reference 18
- 108. Reference 8
- 109. References 17, 18
- 110. Reference 41
- 111. Reference 15
- 112. Reference 16
- 113. U. Khanal , Phys. Rev. D 28 , 1291 (1983)
- 114. Reference 26
- 115. Reference 24
- 116. Reference 15

-: 100 :-

-: 101 :-

- 117. Reference 16
- 118. Reference 14
- 119. Reference 13
- 120. Reference 97
- 121. References 98, 99
- 122. Reference 18
- 123. References 15, 16
- 124. Reference 14
- 125. Reference 13
- 126. References 15, 16
- 127. References 5, 6; S.W. Hawking, Hawking on the Big Bang and Black Holes (World Scientific, 1992)
- 128. Reference 8
- 129. References 5,6
- 130. Reference 8
- 131. References 5, 6
- 132. Reference 8
- 133. References 98, 99
- 134. Reference 19
- 135. Reference 22 , P. 220
- 136. Reference 18
- 137. Reference 15
- 138. Reference 113
- 139. Reference 26
- 140. Reference 24
- 141. Reference 97
- 142. References 98, 99

143. M. Ahmed and A.K. Mondal, Submitted for publication in " GANIT "

- 144. Reference 18
- 145. References 15, 16

146. M. Ahmed and A.K. Mondal, Submitted Revised manuscript in Physics Letters A

- 147. Reference 17
- 148. Reference 14
- 149. Reference 13
- 150. Reference 97
- 151. Reference 143
- 152. References 15 , 16 , 18
- 153. References 13 19, 143, 146
- 154. B.S. Kay and R.M. Wald , Physics Reports 207 , 49 (1991)

Rajshabi University Library Document No D - 2332Date 24.10.04